

**Crawford School of Public Policy** 



**Centre for Applied Macroeconomic Analysis** 

# Germany's Macroeconomic Drivers Through the COVID-19 Pandemic and Recovery Period

# CAMA Working Paper 48/2024 July 2024

## Stefan Hohberger

Munich University of Applied Sciences Centre for Applied Macroeconomic Analysis, ANU

# Abstract

This paper estimates a three-region structural macroeconomic model to analyse the main drivers of GDP, inflation, and wage growth through the COVID-19 pandemic and recovery period in Germany. By incorporating COVID-related shocks, trade in commodities, and endogenous ELB periods, the estimation results suggest: (i) the COVID-19 pandemic in 2020-21 was mainly driven by domestic and foreign lockdown shocks (demand-driven), (ii) the inflation surge in 2021-22 was characterised by an increase in commodity prices, a recovery of global demand, and pronounced supply-side factors, and (iii) wage growth per hour was counterbalanced by competing demand and supply-side effects. Key estimated shocks in the model closely match off-model indicators, supporting its empirical plausibility.

# Keywords

COVID-19, business cycle, inflation, open-economy DSGE model, Bayesian estimation, Germany

## **JEL Classification**

C51, E32, E52, F41, F45

## Address for correspondence:

(E) cama.admin@anu.edu.au

# **ISSN 2206-0332**

<u>The Centre for Applied Macroeconomic Analysis</u> in the Crawford School of Public Policy has been established to build strong links between professional macroeconomists. It provides a forum for quality macroeconomic research and discussion of policy issues between academia, government and the private sector.

**The Crawford School of Public Policy** is the Australian National University's public policy school, serving and influencing Australia, Asia and the Pacific through advanced policy research, graduate and executive education, and policy impact.

# Germany's macroeconomic drivers through the COVID-19 pandemic and recovery period

Stefan Hohberger<sup>a,b</sup>

<sup>a</sup>HM Business School, Munich University of Applied Sciences, Munich, Germany <sup>b</sup>Centre for Applied Macroeconomic Analysis (CAMA), Australian National University, Canberra, Australia

 $17\mathrm{th}$ July2024

#### Abstract

This paper estimates a three-region structural macroeconomic model to analyse the main drivers of GDP, inflation, and wage growth through the COVID-19 pandemic and recovery period in Germany. By incorporating COVID-related shocks, trade in commodities, and endogenous ELB periods, the estimation results suggest: (i) the COVID-19 pandemic in 2020-21 was mainly driven by domestic and foreign lockdown shocks (demand-driven), (ii) the inflation surge in 2021-22 was characterised by an increase in commodity prices, a recovery of global demand, and pronounced supply-side factors, and (iii) wage growth per hour was counterbalanced by competing demand and supply-side effects. Key estimated shocks in the model closely match off-model indicators, supporting its empirical plausibility.

**Keywords:** COVID-19, business cycle, inflation, open-economy DSGE model, Bayesian estimation, Germany

**JEL classification:** C51, E32, E52, F41, F45

<sup>\*</sup>Acknowledgements: I would particularly like to thank Marco Ratto and my former colleagues from the European Commission's Joint Research Centre in Ispra for all the discussions and work on the Global Multi-country (GM) model. I would also like to thank Warwick McKibbin for his helpful comments.

<sup>\*\*</sup>Email: stefan.hohberger@hm.edu

#### 1. Introduction

The COVID-19 pandemic and subsequent energy crisis have led to significant macroeconomic consequences for Germany and the global economy. The immediate response to COVID-19 through the implementation of containment measures and widespread lockdowns, brought many economic activities and international trade to a halt. As a consequence, heightened uncertainties, disrupted global supply chains, and a sharp decline of domestic activity accelerated the economic downturn, leading to a synchronised global recession in 2020. Simultaneously, governments rolled out unprecedented fiscal stimulus packages. The energy crisis in 2021-22 has amplified the surge in inflation, leading to a strong tightening of monetary policy.

This paper analyses the main drivers of the COVID-19 pandemic and recovery period for Germany through the lens of an estimated structural macroeconomic model. The model is a three-region dynamic stochastic general equilibrium (DSGE) model, consisting of Germany (DE), the rest of the euro area (REA), and the rest of the world (RoW). The structural setup builds on Albonico et al. (2019), Hohberger et al. (2020), and Cardani et al. (2022b). Given the focus on the COVID-19 and post-pandemic period, the model has been extended in two directions: First, it incorporates COVID-related shocks along the dimensions of Cardani et al. (2022a, 2023), and second, it extends the use of commodities in the model following the approaches by Giovannini et al. (2019) and Cardani et al. (2023).

The model is estimated using data from the period 1999q1–2023q4, thereby encompassing the COVID-19 pandemic and recovery period. The methodological approach follows Cardani et al. (2022a, 2023) in which the model allows for heteroskedastic exogenous disturbances that only occur during the COVID-period (2020q1-2022q4). Incorporating a large number of time series information (observables) in the estimation process involves a large number of shocks, but allows to assess the main drivers of macroeconomic variables and to provide a plausible narrative of economic developments. Therefore, the model includes: (i) domestic and foreign demand and supply shocks, including private domestic demand, monetary and fiscal policy shocks, shocks to labour and goods market adjustment and productivity, as well as shocks to foreign activity, trade and commodity prices, and (ii) additional COVID-specific macroeconomic disturbances, such as transitory lockdown shocks (forced savings), shocks to the labour market (labour hoarding), fiscal stimulus and foreign 'risk' shocks. Additionally, (iii) the model includes an endogenous and occasionally binding effective lower bound (ELB) on nominal short-term interest rates as in Hohberger et al. (2019) and Croitorov et al. (2020).

The estimation results suggest a crucial role for transitory lockdown shocks (forced savings) in explaining the contraction and recovery of economic activity in Germany during 2020–21, with significant stabilising effects from fiscal policy measures. Spillovers from global demand and supply shocks also affect Germany's economic activity. The GDP recovery period in 2021-23 indicates offsetting effects between a normalisation of domestic and foreign demand and a slowdown in international trade, coupled with adverse supply shocks. This reflects increasing supply-chain bottlenecks for the German economy. The driving factors behind the surge in Germany's CPI inflation in 2021-22 are primarily attributed to domestic and foreign supply-side factors and rising commodity prices. Adverse productivity and supply shocks (price markup and export price shocks) highlight global supply-chain and capacity constraints. Unlike the global financial crisis (GFC), the inflation dynamics during the 2021-22 surge have been driven particularly by supply-side factors. The estimated shocks in the model closely align with off-model indicators that are not part of the observed data, supporting the empirical plausibility of the identified shocks.

The remainder of the paper proceeds as follows. Section 2 presents stylised macroeconomic facts for the German economy since the onset of the GFC in 2008. Section 3 discusses recent studies with respect to modelling COVID-specific developments and empirical literature on the recent post-pandemic surge in inflation. Section 4 outlines the structure of the model. Section 5 describes the model solution, data, and estimation methodology, and discusses posterior estimates. Section 6 discusses the dynamic responses to shocks characteristic of the COVID-19 period. Section 7 quantifies the main drivers of GDP growth, CPI inflation, and wage growth in Germany. Section 8 compares model-implied results with off-model evidence. Section 9 provides a sensitivity analysis without COVID-specific shocks. Section 10 summarises and concludes the paper.

#### 2. Stylised facts

Figure 1 summarises several stylised facts for selected macroeconomic variables during the GFC and the COVID-19 pandemic in Germany. First, the V-shape contraction of economic activity in 2020q2 was more pronounced compared to the more persistent U-shaped recession of the GFC (Figure 1a). Unlike the 2008-09 recession, both private consumption and investment dropped simultaneously during the pandemic in 2020, whereas the GFC recession exhibited a relatively mild downturn in consumption compared to a significant drop in investment growth in 2009.

Second, in both recessions the number of employees remained relatively stable compared to the decline in hours worked (Figure 1b). The wedge between hours worked and employment during the onset of the COVID-pandemic is attributable to the introduction of job retention schemes (short-time work) aimed at moderating employment losses compared to the sharp decline in GDP.



Figure 1: Stylised facts from GFC to COVID-19 in Germany.

Note: Real demand components (inflation rates) are shown in percentage-point deviations from the steady state, which is calibrated to 1.35% (2%) per year, respectively. 1 on the y-axes correspond to 1 pp. Sources: Eurostat and Comext data.

Third, Figure 1c shows that CPI and GDP inflation (lhs) increased significantly from 2021 onward, peaking at 8% and 6.8% in 2022q3 and 2023q2, respectively, when adding the 2% steady-state. The co-movement between GDP and inflation during the 2020 lockdown, similar to the 2008-09 recession, suggests a stronger initial demand-side driver (positive

correlation). The prices of commodities (rhs) declined at the start of the pandemic, as during the GFC, and surged in 2021–22 due to the pandemic recovery, global supply chain bottlenecks, geopolitical tensions, and ultimately, Russia's war of aggression against Ukraine. In 2023, oil and energy prices dropped significantly to pre-crisis levels, also contributing to the decline in CPI inflation.

#### 3. Related literature

This paper analyses the macroeconomic drivers of the COVID-19 pandemic and the recovery period in Germany, and compares them with the GFC recession in 2008-09. It relates broadly to three strands of literature.

The first strand relates to the class of estimated structural macroeconomic models for EMU Member States, with a focus on Germany. Several studies have focused on the macroeconomic drivers of Germany's trade balance since the introduction of the euro (e.g. Kollmann et al., 2015; Hohberger et al., 2020; Cardani et al., 2022b). Gadatsch et al. (2016) analyse the effects of fiscal policy measures in Germany during the GFC, whereas Albonico et al. (2019) provide an extensive cross-country comparison of the macroeconomic dynamics among the four largest EMU Member countries during and after the GFC. Regarding COVID-related studies, Funke and Terasa (2022) examines the unconventional fiscal policy measure of a temporary VAT rate cut during 2020q3-20q4 in a calibrated DSGE model for Germany. They find that the tax policy implied a real GDP increase of about 0.3 percentage points (pp) for 2020. Hinterlang et al. (2023) simulate the German fiscal stimulus packages using a calibrated New Keynesian multi-sector general equilibrium model. They find cumulative short-term stabilising effects on output of up to 6 pp over 2020-22.

The second strand relates to the recent literature that adapts state-of-the art DSGE models to COVID-19 dynamics in order to analyse the drivers of the COVID-19 recession in 2020. For the US, for example, Chen et al. (2020) extends the New York Fed DSGE model by incorporating COVID-specific demand and supply shocks, concluding that the pandemic recession has been rather a demand shock to the US economy. Corrado et al. (2021) estimate a two-sector New Keynesian model on US data and find that the pandemic-induced economic downturn can be explained by a combination of large demand and supply shocks. For the EA, Kollmann (2021) uses a stylised New Keynesian model and argues that the aggregate supply shock has been the main driver of the sharp GDP contraction in the EA in 2020, whereas offsetting demand and supply shocks account for the stability of EA inflation.<sup>1</sup> Cardani et al.

<sup>&</sup>lt;sup>1</sup>There is also a growing literature on structural models that incorporate both epidemiological and economic dynamics to analyse the effects of the pandemic and containment policies (e.g. Eichenbaum et al.,

(2022a) augment and estimate the European Commission's DSGE model for the EA with COVID-specific shocks and financially-constrained investors. They find that 'forced savings' are crucial to explain quarterly real GDP growth during the 2020 recession period. They also provide substantial comparisons to alternative model specifications. Cardani et al. (2023) provide a model-based comparison of the COVID-related recession period between the EA and the US.

Third, a rapidly growing number of studies have emerged recently on the economic developments during and after the pandemic, with a particular focus on the inflation dynamics in the EA and the US, based on various empirical approaches. For the US, the main drivers of the surge in inflation during 2021-2022 are attributed to a combination of binding supply chain constraints (Comin et al., 2023; di Giovanni, 2022; Blanchard and Bernanke, 2023), fiscal stimulus (di Giovanni et al., 2023; Bayer et al., 2023; Jorda and Nechio, 2023), rising commodity prices combined with expansionary policies (Gagliardone and Gertler, 2023; Reis, 2022; Blanchard and Bernanke, 2023), and tight labour markets (Ball et al., 2022) that point to a non-linear Phillips curve in explaining the surge of post-pandemic inflation (Benigno and Eggertsson, 2023, 2024; Harding et al., 2023). In the EA, Neri et al. (2023) and Pasimeni (2022) find that the increase in commodity prices is one of the main drivers in explaining EA inflation during the pandemic, accounting for more than half of the post-COVID headline inflation. Pasimeni (2022) stresses that price pressure stems mainly from sectors with high import content, suggesting the importance of international supply chain disruptions. Hansen et al. (2023) emphasise the crucial role of import prices, accounting for 40% of the EA's change in the consumption deflator in 2022. Cardani et al. (2023) suggest that demand factors play a larger role in US inflation, while supply factors are more significant for EA inflation. They also highlight the importance of rising commodity prices for the surge in inflation during 2021–22.

Blanchard and Bernanke (2023) estimate a dynamic model of prices, wages, and inflation expectations to analyse the direct and indirect effects of product and labour market shocks on US pandemic-era inflation and wage growth. They find price shocks, i.e. commodity prices, sectoral demand shifts, and supply constraints as main sources of inflation. Several studies apply the approach of Blanchard and Bernanke (2023) to EMU Member States, namely Menz (2024) for Germany, Pisani and Tagliabracci (2024) for Italy, and Ghomi et al. (2024) for Spain. Their common findings suggest that the surge in inflation in these EMU Member States has mainly been driven by commodity price shocks and supply bottlenecks, without major evidence of wage-price spirals. Arce et al. (2024) attribute similar drivers to

<sup>2021, 2022).</sup> However, the model in this paper does not include an epidemiology block.

the EA aggregate inflation surge, namely labour market tightness, productivity and global supply chain disruptions, as well as energy and food price shocks.

This paper contributes to the existing literature along several dimensions: (i) It estimates a state-of-the-art large-scale multi-country DSGE model for the German economy (DE-REA-RoW) using data from 1999q1-2023q4; (ii) it incorporates heteroskedastic COVIDrelated domestic and foreign lockdown shocks (transitory saving and investment shocks), shocks to labour demand (labour hoarding) to account for the wedge between hours worked and employees (intensive margin) due to short-time work arrangements, and the use of commodities in production and final consumption demand; (iii) it allows for endogenous, occasionally binding constraints to account for ELB periods; (iv) it uses many time series data (observables) to capture a large number of possible domestic and foreign demand and supply-side drivers of the macroeconomic developments in Germany; (v) it provides off-model evidence for the fit of the model-implied estimated shock pattern.

#### 4. The model

This paper uses a state-of-the-art three-region macroeconomic DSGE model, consisting of Germany (DE), the rest of the euro area (REA), and the rest of the world (RoW). The structural setup builds upon the model used in Hohberger et al. (2020) and Cardani et al. (2022b), but extends the model with (i) COVID-specific shocks following the approach by Cardani et al. (2022a), and (ii) the use of commodities in production and final consumption demand (Cardani et al., 2023).

Concerning the COVID-specific shocks, the following additional exogenous disturbances have been incorporated: (1) A transitory lockdown shock (forced savings),  $\varepsilon_t^{tC}$ , to capture the lockdown-imposed drop in consumption, (2) a transitory labour demand shock (labour hoarding),  $\varepsilon_t^{tN}$ , to capture short-time work arrangements, i.e. to distinguish between hours worked and hours paid, (3) a transitory VAT shock,  $\varepsilon_t^{tvat}$ , to capture the reduction of the German VAT consumption tax during 2020q3-2020q4, and (4) a transitory shock to the investment risk premium,  $\varepsilon_t^{tS}$ , to capture the lockdown-imposed drop in investment demand. To capture COVID-specific demand contraction in the simplified REA and RoW model blocks, the model also incorporates additional foreign shocks to the time preference (savings shock).

The economy of DE consists of various sectors including households, firms that operate domestically or in the import-export sector, as well as a government and a central bank. In contrast, the regional blocks of REA and RoW have a simpler structure. Within the DE block, there is a distinction between two types of households: Ricardian and liquidityconstrained (LC). Ricardian households are characterised by their access to financial markets, ability to smooth consumption, and ownership of firms through equity. On the other hand, LC households lack access to financial markets and consume all their disposable wage and transfer income each period. Both household types supply labour to domestic firms at a common wage established by a labour union with monopoly power. The REA and RoW blocks only include Ricardian households, who supply labour inelastically.

In the domestic production sector, firms operate under monopolistic competition and produce a variety of differentiated intermediate goods. These intermediate goods are then aggregated into domestic value added by perfectly competitive firms. In the subsequent stage, total domestic output is produced by perfectly competitive firms by combining the domestic value added with commodities. The RoW region is the sole producer of commodities, which includes both energy and non-energy commodities. In the import sector, perfectly competitive firms, known as import retailers, purchase goods from foreign regions and assemble them into a final import good. These final import goods are then combined with domestic output by final good packagers to create final aggregate demand component goods.

The DE government is involved in purchasing final goods and providing lump-sum transfers to households. To finance its expenditure, the government issues debt and imposes various taxes including distortionary taxes on labour, capital, and consumption, along with non-distortionary lump-sum taxes. In contrast, the simplified REA and RoW blocks do not contain any fiscal authority. The monetary authorities, however, set short-term nominal interest rates by following a Taylor rule, which reacts to inflation and the output gap. The following description highlights the primary aspects of the model, with further details available in Appendix A.

#### 4.1. Households

There is a continuum of households, indexed by  $j \in [0, 1]$ , whereas a share of households (Ricardians  $\omega^s$ ) owns firms and trades assets. The remaining share  $(1-\omega^s)$  is liquidityconstrained (c) and consumes its entire disposable income in each period ('hand-to-mouth'). Household preferences are defined over consumption and leisure. Additionally, Ricardian utility is determined by the holdings of financial assets.

#### 4.1.1. Ricardian households

Ricardian preferences are given by the infinite horizon expected life-time utility:

$$U_{j}^{s} = E_{0} \sum_{t=0}^{\infty} (\tilde{\beta}_{t})^{t} u_{j,t}^{s}(.), \qquad (1)$$

where  $\tilde{\beta}_t$  is the stochastic discount factor.<sup>2</sup> Ricardian households have full access to financial markets, which allows them to accumulate wealth,  $A_{j,t}$ , consisting of domestic private risk-free bonds,  $B_{j,t}^{rf}$ , domestic government bonds,  $B_{j,t}^G$ , one internationally traded bond,  $B_{j,t}^W$ , and internationally traded shares,  $P_t^S S_{j,t}$ :

$$A_{j,t} = B_{j,t}^{rf} + B_{j,t}^G + e_{RoW,t}B_{j,t}^W + P_t^S S_{j,t}$$
(2)

where  $P_t^S$  is the nominal price of shares. The international bond is issued and denominated in foreign currency, therefore, the financial wealth in terms of domestic currency is also influenced by the nominal exchange rate,  $e_{RoW,t}$ .

Ricardian households gain utility from consumption,  $C_{j,t}^s$ , and experience disutility from labour,  $N_{j,t}^s$ , as well as from holding risky financial assets,  $U_{j,t-1}^A$ . The instantaneous utility function of savers,  $u^s(.)$ , is defined as:

$$u_{j,t}^{s}(C_{j,t}^{s}, N_{j,t}^{s}, \frac{U_{j,t-1}^{A}}{P_{t}^{C,vat}}) = \frac{(C_{j,t}^{s} - \varepsilon_{t}^{tC} - h(C_{t-1}^{s} - \varepsilon_{t-1}^{tC}))^{1-\theta}}{1-\theta} - \frac{\omega^{N} \varepsilon_{t}^{U}(C_{t})^{1-\theta} (N_{j,t}^{s} + \varepsilon_{t}^{tN})^{1+\theta^{N}}}{1+\theta^{N}} - (C_{j,t}^{s} - \varepsilon_{t}^{tC} - h(C_{t-1}^{s} - \varepsilon_{t-1}^{tC}))^{-\theta} \frac{U_{j,t-1}^{A}}{P_{t}^{C,vat}},$$
(3)

where  $C_t^s = \int_0^1 C_{j,t}^s dj$ , *h* measures the strength of external habits in consumption, and  $\omega^N$  is the stochastic weight of the disutility of labour.  $\varepsilon_t^U$  captures a labour supply shock.  $\varepsilon_t^{tC}$  captures the non-persistent lockdown shock (forced saving) that constrain consumption outside of habit persistence,  $\varepsilon_t^{tN}$  captures a labour hoarding shock as in (Cardani et al., 2022a).<sup>3</sup> The disutility of holding risky financial assets,  $U_{j,t-1}^A$ , takes the following form:

$$U_{j,t-1}^{A} = \left(\alpha^{b_{0}} + \varepsilon_{t-1}^{B}\right) B_{j,t-1}^{G} + \left(\alpha^{bw_{0}} + \varepsilon_{t-1}^{bw}\right) e_{RoW,t} B_{j,t-1}^{W} + \frac{\alpha^{bw_{1}}}{2} \frac{(e_{RoW,t-1}NFA_{t-1})^{2}}{P_{t-1}^{Y}Y_{t-1}} + \left(\alpha^{S_{0}} + \varepsilon_{t-1}^{S}\right) P_{t-1}^{S} S_{j,t-1}.$$
(4)

Internationally traded bonds are subject to transaction costs which are a function of the average NFA position relative to GDP. The asset-specific risk premium depends on an asset-specific exogenous shock  $\varepsilon^x$ ,  $x \in \{B, S, bw\}$ , and an asset-specific intercept  $\alpha^x$ ,  $x \in$ 

 $<sup>{}^{2}\</sup>tilde{\beta}_{t} = \beta \exp(\varepsilon_{t-1}^{c})$  features a shock to the subjective rate of time preference (saving shock)  $\varepsilon_{t}^{c}$ .

<sup>&</sup>lt;sup>3</sup>Aggregate consumption,  $C_t$ , in the second term of the right-hand side is introduced as normalisation to ensure a balanced steady-state growth path.

 $\{b_0, S_0, bw_0\}$ . By incorporating a disutility for holding risky assets, the model reflects households' preference for safe assets, such as risk-free short-term bonds. This preference creates an endogenous gap between the returns on risky assets and those on safe bonds (Albonico et al., 2019).

The  $j^{th}$  Ricardian household faces the following budget constraint:

$$P_{t}^{C,vat}C_{j,t}^{s} + A_{j,t} = (1 - \tau^{N})W_{t}(N_{j,t}^{s} + \varepsilon_{t}^{tN}) + (1 + i_{t-1}^{rf})B_{j,t-1}^{rf} + (1 + i_{t-1}^{G})B_{j,t-1}^{G} + (P_{t}^{S} + P_{t}^{Y}D_{t})S_{j,t-1} + (1 + i_{t-1}^{W})e_{RoW,t}B_{j,t-1}^{W}$$

$$+ T_{j,t}^{s} - tax_{j,t}^{s},$$
(5)

where  $P_t^{C,vat}$  is the private consumption deflator<sup>4</sup>,  $W_t$  denotes the nominal wage rate,  $N_{j,t}^s$  is the employment in hours,  $T_{j,t}^s$  are government transfers and  $tax_{j,t}^s$  lump-sum taxes paid by savers.  $i_t^{rf}$ ,  $i_t^G$ , and  $i_t^W$  are returns on domestic private risk-free bonds, domestic government bonds, and internationally traded bonds, respectively. Transfers include unemployment benefits,  $BEN_{j,t}^s$ , defined as the gap between actual and potential hours multiplied with benefit replacement rate,  $\tau^u$ :

$$T_{j,t}^s = BEN_{j,t}^s + \omega^s P_t T_t, \tag{6}$$

$$BEN_{j,t}^{s} = \tau^{u}W_t \Big( N_t^{pot} - (N_{j,t}^{s} + \varepsilon_t^{tN}) \Big).$$
<sup>(7)</sup>

Ricardian households receive nominal profits in form of dividends,  $D_t$ . Gross nominal return on shares  $S_t$  is defined as:

$$1 + i_t^S = \frac{P_t^S + P_t^Y D_t}{P_{t-1}^S}.$$
(8)

Ricardian households maximise the present value of the expected stream of future utility by choosing the amount of consumption,  $C_{j,t}^s$ , and next period asset holdings,  $B_{j,t}^{rf}$ ,  $B_{j,t}^G$ ,  $S_{j,t}$ ,  $B_{j,t}^W$ , subject to their budget constraint (Eq. 5), The optimality conditions can be found in Appendix A.1.

 $<sup>\</sup>overline{{}^{4}P_{t}^{C,vat}}$  is the VAT adjusted private consumption deflator,  $P_{t}^{C,vat} = (1 + \tau^{C} + \varepsilon_{t}^{tVAT})P_{t}^{C}$ , where  $\tau^{C}$  is the tax rate on consumption (VAT) and  $\varepsilon_{t}^{tVAT}$  captures the VAT-tax cut during 2020q3-2020q4 implemented by the German government.

#### 4.1.2. Liquidity-constrained households

Liquidity-constrained (LC) households do not have access to financial markets. Their instantaneous utility function,  $u^{c}(.)$ , is:

$$u_{j,t}^{c}(C_{j,t}^{c}, N_{j,t}^{c}) = \frac{(C_{j,t}^{c} - \varepsilon_{t}^{tC} - h(C_{t-1}^{c} - \varepsilon_{t-1}^{tC}))^{1-\theta}}{1-\theta} - \frac{\omega^{N} \varepsilon_{t}^{U}(C_{t})^{1-\theta}}{1+\theta^{N}} (N_{j,t}^{c} + \varepsilon_{t}^{tN})^{1+\theta^{N}}.$$
(9)

In each time period, they consume their entire net disposable income, which consists of after-tax (paid) labour income and lump-sum transfers from the government:

$$P_t^{C,vat}C_{j,t}^c = (1-\tau^N)W_t(N_{j,t}^c + \varepsilon_t^{tN}) + T_{j,t}^c - tax_{j,t}^c + P^{C,vat}\left(\varepsilon_t^{tC} - \frac{1}{6}\sum_{i=8}^{13}\varepsilon_{t-i}^{tC}\right).$$
(10)

During the COVID-19 pandemic, this constraint is eased such that even LC households can save (forced savings),  $\varepsilon_t^{tC}$ , which will be gradually spent post-pandemic.

#### 4.1.3. Wage setting

Households supply differentiated labour services,  $N_{j,t}^r$ , with  $r = \{s, c\}$ , in a monopolistically competitive labour market. A labour union aggregates labour hours from both types of households into a unified labour service and supplies it to the intermediate goods producing firms. Given the same labour demand schedule, each household works the same average number of hours. The union maximises the weighted average of the members' discounted future utility stream with respect to the wage, subject to the combined budget constraints of the households and the demand for differentiated labour by intermediate-goods producers.

Nominal wage rigidity takes the form of quadratic adjustment costs for changing nominal wages according to  $adj_t^w = \frac{\gamma^w(\sigma^n-1)}{2}W_tN_t(\pi_t^w - \pi^w)^2$ , where  $\sigma^n$  is the inverse of the steady gross wage markup and  $\pi_t^w$  is the wage inflation. Real wage rigidity is modelled in the spirit of Blanchard and Galí (2007), implying a gradual adjustment of past reals wages to changes in the price level. The wage rule is determined by equating the marginal utility of leisure,  $U_t^N$ , to the weighted average of the marginal utility of consumption,  $\lambda_t$ , times the real wage adjusted for a wage markup. The wage equation is:

$$\mu^{w} \left[ \frac{U_{t}^{N}}{\lambda_{t}} - \tau^{u} \frac{W_{t}}{P_{t}^{C,vat}} - \mu_{t}^{w} \right]^{1-\gamma^{wr}} \left[ (1-\tau^{N}) \frac{W_{t-1}}{P_{t-1}^{C,vat}} \right]^{\gamma^{wr}} = (1-\tau^{N}) \frac{W_{t}}{P_{t}^{C,vat}}, \tag{11}$$

where  $\mu_t^w$  is the cyclical gross wage markup:

$$\mu_t^w = \gamma^w \left[ \frac{\partial a dj_t^w}{\partial W_t} - \beta E_t \frac{\lambda_{t+1}}{\lambda_t} \frac{1}{\pi_{t+1}^{C,vat} + 1} \frac{\partial a dj_{t+1}^w}{\partial W_t} \right].$$
(12)

 $\mu^w$  is the steady state gross wage markup,  $\gamma^w$  and  $\gamma^{wr}$  represent the degree of nominal and real wage rigidity, respectively, and  $\varepsilon_t^U$  captures a labour supply shock (wage markup). The marginal utility of leisure is defined as:  $U_t^N = \omega^N (C_t)^{1-\theta} (N_t)^{-\theta^N}$ , and  $\lambda_t$  the weighted average of the marginal utility of consumption.

#### 4.2. Production sector

The total domestic output is a combination of domestic value added and commodities. Value added consists of a bundle of differentiated goods produced by monopolistically competitive firms, which utilise capital and labour in their production processes.

#### 4.2.1. Total output demand

Perfectly competitive firms produce total output,  $O_t$ , by combining value added,  $Y_t$ , with (imported) commodities,  $CO_t$ , using the following CES production function:

$$O_t = \left[ \left( 1 - s^{CO} \exp(\varepsilon_t^{CO}) \right)^{\frac{1}{\sigma^o}} (Y_t)^{\frac{\sigma^o - 1}{\sigma^o}} + \left( s^{CO} \exp(\varepsilon_t^{CO})^{\frac{1}{\sigma^o}} (CO_t)^{\frac{\sigma^o - 1}{\sigma^o}} \right]^{\frac{\sigma^o}{\sigma^o - 1}}.$$
 (13)

 $s^{CO}$  represents the commodity input share, influenced by the exogenous process  $\varepsilon_t^{CO}$ .  $\sigma^o$  refers to the elasticity of substitution between factors. Each firm maximises its expected profits:

$$\max_{Y_t, IS_t} P_t^O O_t - P_t^Y Y_t - P_t^{CO} CO_t,$$
(14)

subject to the production function (13). The first order conditions for the demand for value added and commodities are:

$$Y_t = \left(1 - s^{CO} u_t^{CO}\right) \left(\frac{P_t^Y}{P_t^O}\right)^{-\sigma^o} O_t,\tag{15}$$

$$CO_t = s^{CO} u_t^{CO} \left(\frac{P_t^{CO}}{P_t^O}\right)^{-\sigma^o} O_t.$$
(16)

Commodities are assumed to be produced only by RoW. Consequently, all commodities used by Germany are imported from RoW, and their price is taken as given:

$$P_t^{CO} = e_{Row,t} P_{Row,t}^{CO} + \tau^{CO} P_t^{Y0}, \tag{17}$$

where  $e_{Row,t}$  is the exchange rate,  $\tau^{CO}$  and  $P^{Y0}$  are the excise duty and the (global) GDP trend deflator, respectively. The total output aggregate price index is given by:

$$P_t^O = \left[ (1 - s^{CO} u_t^{CO}) (P_t^Y)^{\sigma^o - 1} + s^{CO} u_t^{CO} (P_t^{CO})^{\sigma^o - 1} \right]^{\frac{1}{1 - \sigma^o}}.$$
 (18)

#### 4.2.2. Value added and intermediate goods producers

Value added,  $Y_t$ , is produced by firms operating in a perfectly competitive market. These firms combine a variety of differentiated goods,  $Y_{i,t}$ , which are produced by firms in a monopolistically competitive market. The differentiated goods are produced using a Cobb-Douglas production function with labour,  $N_{i,t}$ , private capital,  $K_{i,t-1}$ , and public capital,  $K_{i,t-1}^G$ , as input factors:

$$Y_{i,t} = \left[A_t^Y \left( (N_{i,t} + \varepsilon_t^{tN}) - FN \right) \right]^{\alpha} (CU_{i,t}K_{i,t-1})^{1-\alpha} (K_{t-1}^G)^{1-\alpha_G} - A_t^Y FC_i.$$
(19)

 $\alpha$  is the steady state labour share,  $A_t^Y$  is an exogenous common labour-augmenting stochastic productivity scalar,  $CU_{i,t}$  and FN are firm-specific levels of capacity utilisation and labour overhead, respectively.  $FC_i$  represents fixed costs in production. Gross investment,  $I_{i,t}$ , drives the law of motion for private capital  $K_{i,t} = I_{i,t} + (1 - \delta)K_{i,t-1}$ , with the depreciation rate  $\delta$ . Public capital,  $K_{i,t-1}^G$ , follows an analogous accumulation equation with output elasticity  $\alpha_G$ .  $A_t^Y$  is a non-stationary process with two types of technology shocks,  $\varepsilon_t^A$  and  $\varepsilon_t^{GA}$ . They are related to a non-stationary process and its autoregressive component  $\rho^A$ :

$$log(A_t^Y) - log(A_{t-1}^Y) = g_t^A + \varepsilon_t^A,$$
(20)

$$g_t^A = \rho^A g_{t-1}^A + (1 - \rho^A) g^A + \varepsilon_t^{GA},$$
(21)

where  $g_t^A g^A$  are the time-varying growth and the long-run growth of technology, respectively.

Following the approach by Cardani et al. (2022a), the model incorporates a transitory labour demand shock (labour hoarding,  $\varepsilon_t^{tN}$ ) to capture short-time work arrangements during the onset of COVID-19, i.e. employees working fewer hours while remaining employed, thereby introducing a wedge between hours worked (production function) and hours paid (wage income). It enters as a transitory shock to hours worked. Dividends are defined as:

$$D_{i,t} = P_{i,t}^{Y} Y_{i,t} - W_t (N_{i,t} + \varepsilon_t^{tN}) - P_t^{I} (I_{i,t} + \varepsilon_t^{tS}) - adj_{i,t},$$
(22)

where  $W_t$  and  $P_{i,t}^I$  are the nominal wage rate and the price of investment goods, respectively.

 $\varepsilon_t^{tS}$  represents an investment-specific lockdown shock. Following Rotemberg (1982), firms face quadratic adjustment costs,  $adj_{i,t}$ , associated with the output price,  $P_{i,t}^Y$ , labour input,  $N_{i,t}$ , private investment,  $I_{i,t}$ , and capacity utilisation,  $CU_{i,t}$ . The adjustment cost definitions and optimality conditions can be found in Appendix A.2.

#### 4.3. International trade

Final good packagers combine domestic output and imported goods to supply different sectors in the economy with components of aggregate demand. Imported goods are a bundle of imports sourced from different origins and are put together by import retailers.

#### Final good packagers

The final aggregate demand-component goods are manufactured by perfectly competitive firms that combine domestic output,  $O_t^{\mathcal{Z}}$ , with imported goods from REA and RoW,  $M_t^{\mathcal{Z}}$ , using a CES production function. The demand for final goods,  $\mathcal{Z} = \{C, G, I, I^G, X\}$ , comes from households and the government, private and public investors, as well as exporters of final goods, respectively:

$$\mathcal{Z}_t = A_t^{p^{\mathcal{Z}}} \left[ (1 - s_t^{M, \mathcal{Z}})^{\frac{1}{\sigma^z}} (O_t^{\mathcal{Z}})^{\frac{\sigma^z - 1}{\sigma^z}} + (s_t^{M, \mathcal{Z}})^{\frac{1}{\sigma^z}} (M_t^{\mathcal{Z}})^{\frac{\sigma^z - 1}{\sigma^z}} \right]^{\frac{\sigma^z}{\sigma^z - 1}},$$
(23)

 $\sigma^{z}$  represents the elasticity of substitution of imports, and  $A_{t}^{p^{z}}$  is a shock to productivity in the sector producing goods,  $\mathcal{Z}$ .  $s_{t}^{M,\mathcal{Z}}$  is a sector-specific stochastic import share, where  $s_{t}^{M,\mathcal{Z}} = exp(\varepsilon_{t}^{M,\mathcal{Z}})s^{M,\mathcal{Z}}$ .  $s^{M,\mathcal{Z}}$  denotes the steady state import share of  $\mathcal{Z}$ . Demand for domestic output and imported goods is given by:

$$O_t^{\mathcal{Z}} = (A_t^{p^{\mathcal{Z}}})^{\sigma^z - 1} \left(1 - s_t^{M, \mathcal{Z}}\right) \left(\frac{P_t^O}{P_t^{\mathcal{Z}}}\right)^{-\sigma^z} \mathcal{Z}_t,$$
(24)

$$M_t^{\mathcal{Z}} = (A_t^{p^{\mathcal{Z}}})^{\sigma^z - 1} s_t^{M, \mathcal{Z}} \left(\frac{P_t^M}{P_t^{\mathcal{Z}}}\right)^{-\sigma^z} \mathcal{Z}_t,$$
(25)

The price deflator associated to the demand components is:

$$P_t^{\mathcal{Z}} = (A_t^{p^{\mathcal{Z}}})^{-1} \left[ (1 - s_t^{M, \mathcal{Z}}) (P_t^O)^{1 - \sigma^z} + s_t^{M, \mathcal{Z}} (P_t^M)^{1 - \sigma^z} \right]^{\frac{1}{1 - \sigma^z}}.$$
 (26)

Total non-commodity imports are defined as:

$$M_t = M_t^C + M_t^I + M_t^G + M_t^{IG} + M_t^X.$$
 (27)

#### Import retailers

Final non-commodity imports are produced by perfectly competitive firms that combine goods from the foreign regions into a final import good. The demand for goods from country l is:

$$M_{l,t} = s_{l,t}^M \left(\frac{P_{l,t}^M}{P_t^M}\right)^{-\sigma^{FM}} M_t \frac{size}{size_l},$$
(28)

and the import prices deflator is:

$$P_t^M = \left[\sum_{l} s_{l,t}^M (P_{l,t}^M)^{1-\sigma^{FM}}\right]^{\frac{1}{1-\sigma^{FM}}}.$$
(29)

 $\sigma^{FM}$  represents the price elasticity of demand for country *l*'s goods, and  $P_{l,t}^M$  is the price of its good. The good from country *l* is purchased at the export price of country *l*,  $P_{l,t}^X$ . Hence, the import price for the domestic country is defined as:

$$P_{l,t}^M = e_{l,t} P_{l,t}^X.$$
 (30)

#### 4.4. Fiscal policy

The government raises taxes on consumption,  $\tau^{C}$ , corporate profits,  $\tau^{K}$ , lump-sum taxes,  $tax_{t}$ , and wage income tax,  $\tau_{t}^{N}$ . It finances consumptive purchases,  $G_{t}$ , investments,  $IG_{t}$ , and transfers,  $T_{t}$ . The tax on commodity imports,  $\tau^{CO}$ , is fixed.  $\tau^{FN}$  denotes a labour hoarding subsidy. Nominal debt evolves as:

$$B_t^G = (1 + i_t^G)B_{t-1}^G - R_t^G + P_tG_t + P_tIG_t + P_tT_t,$$
(31)

where  $R_t^G$  are the nominal government revenues:

$$R_t^G = (\tau^C + \varepsilon_t^{tVAT}) P_t C_t + \tau^K \Big( P_t Y_t - W_t (N_t + \varepsilon_t^{tN}) - \delta P_t K_{t-1} \Big)$$
(32)

$$+\tau_t^N (N_t + \varepsilon_t^{tN}) W_t + \tau^{CO} P_t^{CO} CO_t + \tau^{FN} W_t \Big( (N_t + \varepsilon_t^{tN}) - N_t \Big) + tax_t.$$
(33)

Lump-sum taxes,  $tax_t$ , adjust residually as government budget closure:

$$tax_{t} = \rho_{\tau} tax_{t-1} + \eta^{deft} \left( \frac{\Delta B_{t-1}^{G}}{Y_{t-1}P_{t-1}^{Y}} - DEFTAR \right)$$
$$+ \eta^{BT} \left( \frac{B_{t-1}^{G}}{Y_{t-1}P_{t-1}^{Y}} - BTAR \right) + \varepsilon_{t}^{tax},$$
(34)

where DEFTAR and BTAR are the targets on government deficit and government debt, respectively, and  $\varepsilon_t^{tax}$  captures a shock. The government increases (decreases) taxes when the level of government debt and the government deficit is above (below) the debt and deficit target. On the spending side, government consumption,  $G_t$ , investment,  $IG_t$ , and transfers,  $T_t$  follow AR(1) processes and are subject to idiosyncratic shocks ( $\varepsilon^G$ ,  $\varepsilon^{IG}$  and  $\varepsilon^T$ ).

#### 4.5. Monetary policy

Monetary policy follows a Taylor (1993)-type policy rule subject to an ELB constraint. The target interest rate,  $i_{EA,t}^{not}$ , responds sluggishly to (quarterly annualised) deviations of EA-wide CPI inflation ( $\pi_{EA,t}^{C,QA}$ ) and EA output gap ( $\hat{Y}_{EA,t}^{QA}$ ) from their respective target levels:<sup>5</sup>

$$i_{EA,t}^{not} - \bar{i} = \rho_{EA}^{i} (i_{EA,t-1} - \bar{i}) + (1 - \rho_{EA}^{i}) \left[ \eta_{EA}^{i\pi} 0.25 \left( \pi_{EA,t}^{C,QA} - \bar{\pi}_{EA}^{C,QA} \right) + \eta_{EA}^{iy} \hat{Y}_{EA,t}^{QA} \right].$$
(35)

 $\bar{i} = \bar{r} + \bar{\pi}^{Yobs}$  is the steady state nominal interest rate, which equals the sum of the steady state real interest rate and trend inflation. The policy parameters  $(\rho^i, \eta^{i\pi}, \eta^{iy})$  capture interest rate inertia and the response to the EA inflation and EA output gap, respectively.

The effective policy rate,  $i_{EA,t}$ , corresponds to the target nominal short-term rate as long as the latter is above the ELB,  $(i^{LB})$ . The effective policy rate satisfies:

$$i_{EA,t} = max\{i_{EA,t}^{not}, i^{LB}\} + \varepsilon_{EA,t}^{i},$$
(36)

where  $\varepsilon_{EA,t}^{i}$  captures a monetary policy shock. More details on the ELB treatment will be explained in section 5.

#### 4.6. Commodities

Following Cardani et al. (2023), commodities are traded at destination-specific prices, influenced by exogenous supply shocks.<sup>6</sup> The total demand for commodities,  $CO_t$ , comprises household energy consumption,  $C_t^E$ , and the demand for industrial supplies in final goods production,  $IS_t$ . Final household consumption is represented as a CES aggregate of commodities used for consumption and final manufactured goods  $C_t^{FG}$ :

$$C_{t} = \left[ (s_{t}^{E})^{\frac{1}{\sigma^{E}}} (C_{t}^{E})^{\frac{\sigma^{E}-1}{\sigma^{E}}} + (1 - s_{t}^{E})^{\frac{1}{\sigma^{E}}} (C_{t}^{FG})^{\frac{\sigma^{E}-1}{\sigma^{E}}} \right]^{\frac{\sigma^{E}}{\sigma^{E}-1}},$$
(37)

<sup>&</sup>lt;sup>5</sup>Potential output,  $Y_{EA,t}^{pot}$ , is defined as the output level that prevails when labour input equals steady state per capita hours worked, the capital stock is utilised at full capacity, and TFP equals its trend component.

<sup>&</sup>lt;sup>6</sup>Unlike Giovannini et al. (2019), this model incorporate an endogenous supply equation where commodity demand is affecting global commodity prices.

where  $s_t^E$  represents the energy share and  $\sigma^E$  the elasticity of substitution.

#### 4.7. Closing the economy

The resource constraint of the DE economy is:

$$Y_t P_t^Y + \tau^{CO} C O_t P_t^{Y0} = P_t^C C_t + P_t^I I_t + P_t^{IG} I G_t + P_t^G G_t + T B_t,$$
(38)

where  $TB_t$  is the trade balance, defined as the difference between exports and (non-commodity) imports:

$$TB_{t} = P_{t}^{X}X_{t} - \sum_{l} \frac{size_{l}}{size} P_{l,t}^{M}M_{l,t} - P_{RoW,t}^{CO}CO_{RoW,t}e_{RoW,t}.$$
(39)

Domestic holding of the internationally traded bond,  $B_t^W$ , evolves according to:

$$e_{RoW,t}B_t^W = (1+i_{t-1}^{bw})e_{RoW,t}B_{t-1}^W + TB_t + ITRP_t^Y Y_t.$$
(40)

ITR represents international transfers that allow to calibrate a non-zero steady state of the trade balance. The sum of all countries' net foreign assets are zero:

$$\sum_{l} NFA_{l,t} size_{l} = 0.$$
(41)

#### 4.8. The REA and RoW blocks

The REA and RoW (subscript k = REA, RoW) model blocks include a budget constraint for the representative household, demand functions for both domestic and imported goods, a linear production technology, a New Keynesian Phillips curve, and a Taylor rule. Both regions do not take capital accumulation into account. The simplified model blocks are subject to various shocks, including those affecting labour productivity, price markups on final output, the subjective discount rate, the relative preference for domestic versus imported goods, and monetary policy surprises.

The household budget constraint in the REA, as a commodity importer, is defined as:

$$Y_{REA,t}P_{REA,t}^{Y} + \tau^{CO}CO_{REA,t}P^{Y0} = P_{REA,t}^{C}C_{REA,t} + TB_{REA,t},$$
(42)

where  $\tau^{CO}CO_{REA,t}P^{Y0}$  represents the excise duty.

Final aggregate demand,  $C_{k,t}$ , is a combination of domestic output,  $Y_{k,t}^C$ , and imported goods,  $M_{k,t}^C$ , using the following CES function:

$$C_{k,t} = A_{k,t}^{p} \left[ \left(1 - \varepsilon_{k,t}^{M} s_{k}^{M}\right)^{\frac{1}{\sigma_{k}^{c}}} \left(Y_{k,t}^{C}\right)^{\frac{\sigma_{k}^{c} - 1}{\sigma_{k}^{c}}} + \left(\varepsilon_{k,t}^{M} s_{k}^{M}\right)^{\frac{1}{\sigma_{k}^{c}}} \left(M_{k,t}^{C}\right)^{\frac{\sigma_{k}^{c} - 1}{\sigma_{k}^{c}}} \right]^{\frac{\sigma_{k}^{c}}{\sigma_{k}^{c} - 1}}.$$
(43)

 $\sigma_k^c$  represents the import elasticity of substitution,  $A_t^{p^C}$  is a shock to productivity in the sector producing goods, C, and  $s_t^M$  is the import share. The demand for domestic and imported goods is obtained from profit maximisation:

$$Y_{k,t}^{C} = (A_{k,t}^{p^{C}})^{\sigma_{k}^{c}-1} \left(1 - \varepsilon_{k,t}^{M} s_{k}^{M}\right) \left(\frac{P_{k,t}^{Y}}{P_{k,t}^{C}}\right)^{-\sigma_{k}^{c}} C_{k,t},$$
(44)

$$M_{k,t}^{C} = (A_{k,t}^{p^{C}})^{\sigma^{c}-1} \varepsilon_{k,t}^{M} s_{k}^{M} \left(\frac{P_{k,t}^{M}}{P_{k,t}^{C}}\right)^{-\sigma_{k}^{c}} C_{k,t},$$
(45)

where the consumer price deflator,  $P_{k,t}^C$ , is:

$$P_{k,t}^{C} = (A_{k,t}^{p^{C}})^{-1} \left[ (1 - \varepsilon_{k,t}^{M} s_{k}^{M}) (P_{k,t}^{Y})^{1 - \sigma_{k}^{c}} + \varepsilon_{k,t}^{M} s_{k}^{M} (P_{k,t}^{M})^{1 - \sigma_{k}^{c}} \right]^{\frac{1}{1 - \sigma_{k}^{c}}}.$$
(46)

The good producers use labour as input factor,  $Y_{k,t} = A_{k,t}^Y N_{k,t}$ , where  $A_{k,t}^Y$  represents trend productivity. The price setting equation follows a New Keynesian Phillips curve:

$$\pi_{k,t}^Y - \bar{\pi}_k^Y = \beta \frac{\lambda_{k,t+1}}{\lambda_{k,t}} (\pi_{k,t+1}^Y - \bar{\pi}_k^Y) + \phi_k^Y log(\frac{Y_{k,t}}{\bar{Y}_k}) + \varepsilon_{k,t}^Y, \tag{47}$$

where  $\lambda_{k,t} = (C_{k,t} - h_k C_{k,t-1})^{-\theta_k}$  is the marginal utility of consumption, and  $\varepsilon_{k,t}^Y$  is a cost-push shock.

REA and RoW total nominal exports are defined as:  $P_{k,t}^X X_{k,t} = \sum_l P_{l,k,t}^X M_{l,k,t}$ , with the bilateral export price being defined as the domestic price subject to a bilateral price shock,  $P_{l,k,t}^X = exp(\varepsilon_{l,k,t}^X)P_{k,t}^Y$ .

Combining the two region's FOCs with respect to international bonds derive the uncovered interest parity (UIP) condition:

$$E_t \left[ \frac{e_{RoW,EA,t+1}}{e_{RoW,EA,t}} \right] (1 + i_{RoW,t}) = (1 + i_{EA,t}) + \varepsilon_{EA,t}^{bw} + \alpha_{EA}^{bw0} + \alpha_{EA}^{bw1} \frac{e_{RoW,EA,t} B_{EA,t}^w}{P_{EA,t}^Y Y_{EA,t}}, \qquad (48)$$

where  $\varepsilon_{EA,t}^{bw}$  captures a euro exchange rate shock (shock to the bond premium between EA and RoW), and  $\alpha_{EA}^{bw1}$  is a debt-dependent country risk premium on NFA holdings to ensure long-run stability of the model (Schmitt-Grohe and Uribe, 2003).

## 4.9. RoW commodity supply

The RoW exclusively supplies two distinct commodities, namely oil,  $CO^{Oil}$  and non-oil commodities,  $CO^{IS}$ , such as natural gas and materials, to domestic and foreign firms.  $\varepsilon_t^{CO}$  captures exogenous commodity supply shocks. The RoW producer combines oil (*Oil*) and

non-oil (IS) commodities into bundles, CO, that are either exported to DE and REA or used domestically. The price of the commodity bundle is specific to its destination and includes a shock term,  $\varepsilon_{l,t}^{P,CO}$ , where l = (DE, REA), aiming to reflect price variations due to differing commodity baskets. Therefore:

$$P_{l,t}^{CO} = \varepsilon_{l,t}^{P^{CO}} \left[ s_l^{Oil} \left( P_t^{Oil} \right)^{1 - \sigma^{CO}} + (1 - s_l^{Oil}) \left( P_t^{IS} \right)^{1 - \sigma^{CO}} \right]^{\frac{1}{1 - \sigma^{CO}}},\tag{49}$$

$$P_t^{CO} = \left[ s^{Oil} \left( P_t^{Oil} \right)^{1 - \sigma^{CO}} + (1 - s^{Oil}) \left( P_t^{IS} \right)^{1 - \sigma^{CO}} \right]^{\frac{1}{1 - \sigma^{CO}}}.$$
 (50)

Commodity prices are exogenous and follow:

$$P_t^{CO} = \frac{P_t}{A_t^{CO}}, \quad \text{where CO} = (\text{Oil, IS})$$
(51)

where  $A_t^{CO}$  is the exogenous commodity-specific productivity technology.

#### 5. Model solution and econometric approach

#### Model solution

The following non-linear system summarises the observation and state equations of the model:

$$y_t^{obs} = \Psi_1(\Theta)S_t,\tag{52}$$

$$S_t = \Psi_1(\Theta)S_{t-1} + \Theta_{\varepsilon}(\theta)\varepsilon_t, \quad \varepsilon_t \sim N(0, Q_t I)$$
(53)

In observation equation (52),  $y_t^{obs}$  denotes the vector of observables at time t, and  $\Psi_1(\Theta)$  links the model variables to the data. The state equation (53) describes the transition of the system's state variables,  $S_t$ , where  $\Psi_1(\Theta)$  and  $\Theta_{\varepsilon}$  are the coefficient matrices. Following Cardani et al. (2022a), the model shocks,  $\varepsilon_t$ , follow a normal distribution with time-varying covariance matrix  $Q_t I$ , such that the state equation (53) incorporates deterministic hetero-skedasticity:

$$Q_t = \begin{cases} Q^{COVID} & \text{for } t \in \{2020q1 : 2022q4\}, \\ Q & \text{otherwise.} \end{cases}$$
(54)

For the pandemic period 2020q1-2022q4,  $Q_t = Q^{COVID}$  incorporates temporary COVID-specific shocks, whereas prior to COVID-19,  $Q_t = Q$ , implies zero standard deviations (and zero expectations) for these shocks.

#### Estimation procedure and filtering

The estimation proceeds in three steps:

1. A subset of parameters is calibrated to match historical long-run properties, such as steady-state ratios.

2. The remaining parameters are estimated using data over the sample period 1999q1 - 2023q4. The full-sample estimation procedure incorporates the estimation of heteroskedastic COVID-specific variances during the period 2020q1–2022q4, thereby allowing for time-varying shock disturbances  $Q_t$ .<sup>7</sup> The estimation employs a linear Kalman filter and a parallelised slice sampling algorithm to draw parameters from their posterior distribution using Markov Chain Monte Carlo methods.<sup>8</sup> We use the Dynare software to solve the linearised model and to perform the estimation (Adjemian et al., 2024).

3. The model accounts for endogenous effective lower bound (ELB) periods using the OccBin approach by Guerrieri and Iacoviello (2015). More precisely, the paper employs a piecewise linear Kalman filter, as in Giovannini et al. (2021), to identify the structural shocks until 2023q4 given the parameter estimates, accounting for ELB periods.<sup>9</sup>

#### Data

The estimation process employs both quarterly and annual data spanning from 1999q1 to 2023q4. The model is estimated at a quarterly frequency, integrating interpolated annual data for series not available at higher frequencies. The data for Germany and the Euro Area aggregate (EA19) are taken from Eurostat. Bilateral trade flows are derived from trade shares using the GTAP trade matrices, covering both goods and services. The annual data for the RoW are compiled from the IMF International Financial Statistics (IFS) and World Economic Outlook (WEO) databases. The estimated model incorporates 41 observed series and accounts for 41 exogenous shocks. Details on the observed time series can be found in Appendix D. The extensive number of shocks corresponds to the large set of observables

<sup>&</sup>lt;sup>7</sup>Cardani et al. (2022a, 2023) employ a two-step approach, where the parameters are estimated using data only until 2019q4 in a first step. In a second step, they estimate the variances of heteroskedastic COVID-specific shocks using data for the pandemic period (2020q1–2021q4 or 2020q1–2022q4, respectively), while keeping all other parameters unchanged by initialising the system's state and covariance matrix at their estimates from the first step.

<sup>&</sup>lt;sup>8</sup>The slice sampler algorithm was introduced by Neal (2003). Planas et al. (2015) reconsider the slices along the major axis of the ellipse to better fit the distribution than any Euclidean slices. The slice sampler has been shown to be more efficient and to offer better mixing properties than the Metropolis-Hastings sampler (Calés et al., 2017). The slice sampler has been used, e.g., by Giovannini et al. (2019) and Hohberger et al. (2019, 2023).

 $<sup>^{9}</sup>$ Cardani et al. (2023) provides additional methodological details. Similar approaches of retrieving endogenous duration of ELB periods have been used, e.g., by Hohberger et al. (2019) and Croitorov et al. (2020).

utilised for the estimation.

#### Calibrated parameters

Steady state ratios in the model are calibrated to match the average historical data for DE, REA and RoW. The steady state shares of DE, REA, and RoW in world GDP are 5.4%, 14.5%, and 80.1%, respectively. The trade-related parameters, specifically the degree of openness and import preferences, reflect the average import content of demand components as computed by Bussière et al. (2013). For Germany, the steady state ratios of private consumption, investment, and government expenditure to GDP are 55%, 19%, and 21%, respectively. The global trend GDP growth rate and trend inflation rate are 1.25% and 2% per year, respectively. The annual depreciation rate of capital is 5.9%. The rate of time preference is set at 0.25% per quarter. Additionally, the steady state share of Ricardian households is calibrated to be 61%, based on the survey by Dolls et al. (2012). The steady state government debt to annual GDP ratio is 62%. Table B.1 in Appendix B presents an overview of selected calibrated parameters.

#### Posterior estimates

Table 1 reports the prior and posterior estimates, as well as the 90% highest posterior density (HPD) intervals, for selected key model parameters. The estimated EA monetary policy parameters suggest a strong response to EA inflation (2.03) compared to the EA output gap (0.03), together with relatively high interest rate inertia (0.90). Estimated habit persistence is 0.88, implying a slow adjustment of consumption to changes in income. Risk aversion and the inverse labour supply elasticity are 1.48 and 2.92, respectively, and similar to those in the literature (e.g., Hohberger et al., 2020; Cardani et al., 2022b). Concerning international trade estimates, the price elasticity of import demand is 1.02, and the price elasticity of commodity demand is 0.19. The posterior estimates also suggest sticky prices (31.8), wages (25.8), and investments (59.1). The estimated parameters for REA and RoW can be found in Table B.2 in Appendix B. Table B.3 and Table B.4 in Appendix B report the main estimated shock processes for DE, and REA and RoW, respectively.

Table 2 reports the posterior estimates of the pandemic-specific innovations that have specifically affected the COVID-19 period. The estimation suggests a mixture of demandside (lockdown shocks) and supply-side shocks (labour market). More precisely, it indicates a pronounced incidence of consumption-specific lockdown shocks (4.75%). It also suggests substantial labour hoarding shocks (2.57%), in line with the measures and labour market policies adopted by Germany. The estimated VAT-tax cut during the period 2020q3-2020q4 is 1.7%. Given the simplified REA and RoW model structure, the estimation suggests a transitory (lockdown) and persistent savings shock (risk shock) for REA to capture the

		Prior d	istribution	Posterior distribution
		Distr	Mean St.Dev	DE
EA	Monetary	Policy		
Interest rate persistence	$ ho^i$	G	$0.85 \\ 0.05$	0.91
Response to inflation	$\eta^{i,\phi}$	G	2.00	1.78
Response to GDP	$\eta^{i,y}$	G	$0.20 \\ 0.10 \\ 0.04$	$(1.56, 2.19) \\ 0.05 \\ (0.03, 0.11)$
	Preferen	ces		
Consumption habit persistence	h	в	$0.50 \\ 0.20$	0.90 (0.83, 0.92)
Risk aversion	θ	G	1.50	1.33 (1.16, 1.70)
Inverse Frisch elasticity of labour supply	$\theta^N$	G	2.50	3.72
Import price elasticity	$\sigma^{z}$	G	2.00	(2.21, 4.86) 1.93 (1.04, 2.10)
Oil price elasticity	$\sigma^{o}$	G	$0.40 \\ 0.5 \\ 0.1$	(1.64, 2.19) 0.40 (0.32, 0.48)
Nomina	al and rea	al frictio	ns	
	P	~	2.2	10.0
Price adjustment cost	$\gamma^{*}$	G	20 12	40.8 (33.7, 49.9)
Nominal wage adjustment cost	$\gamma^w$	G	20 12	17.3 (9.2, 19.4)
Real wage rigidity	$\gamma^{wr}$	В	$0.47 \\ 0.20$	0.85 (0.54, 0.89)
Employment adjustment cost	$\gamma^N$	G	20 12	1.8 (15.23)
Capacity utilisation quadratic adj cost	$\gamma^{CU,2}$	G	0.003	0.003
Investment adjustment cost	$\gamma^{I,2}$	G		(0.002, 0.005) 38.2 (19.1, 67.5)
	Fiscal pol	licy		
Lump-sum tax persistence	$\rho^{tax}$	в	0.85	0.82
Tax response to deficit	$\eta^{def}$	В	0.06 0.03 0.008	(0.75, 0.89) 0.03 (0.02, 0.04)

#### Table 1: Prior and posterior distribution of key estimated DE model parameters.

Note: Cols. (1)-(2) list model parameters. Cols. (3)-(4) indicate the prior distribution function (B: Beta distribution; G: Gamma distribution). Col. (5) shows the mode and the 90% HPD intervals of the posterior distributions of model parameters.

COVID-19 pattern during 2020 and the recovery in 2021-2022. For the RoW, the estimation suggests an additional persistent savings shock (risk shock) to capture the pattern of the pandemic period.<sup>10</sup>

<sup>&</sup>lt;sup>10</sup>The interpolated annual data for RoW provide a less pronounced contraction of GDP growth in 2020q2, for which the estimation does not suggest proper identification of the consumption-specific lockdown shock as in DE and REA.

		Prior d	istribution	Posterior distribution
		Distr	Mean St.Dev	
DE COVID-specific	shocks (s	tandard	deviations	in %)
Consumption-specific lockdown shock	$\varepsilon^{tC}$	G	5	4.75
			2	(4.03, 7.36)
Investment-specific lockdown shock	$\varepsilon^{tS}$	G	5	2.27
			2	(1.10, 5.80)
labour hoarding shock	$\varepsilon^{tN}$	G	5	2.57
			2	(1.85, 4.77)
VAT tax shock	$\varepsilon^{tVAT}$	G	2	1.70
			0.8	(0.91, 3.03)
REA COVID-specifi	c shocks (	standard	deviation	a in %)
	e shoeks (	Standard		5 m 70)
Consumption-specific lockdown shock	$\varepsilon^{tC}$	G	5	6.97
			2	(4.99, 9.78)
Risk shock	$\varepsilon^{t\beta}$	G	5	8.30
			2	(3.91, 10.13)
RoW COVID-specifi	c shocks (	standard	deviation	s in %)
Diele ek este	$_{-}t\beta$	C	F	19.76
RISK SHOCK	8.00	G	ວ ວ	13.70
			2	(12.03, 17.44)

Table 2: Prior and posterior distribution of estimated COVID-specific innovations.

Note: This table reports the mode and the standard deviation (in %) of the posterior distributions of DE, REA and RoW COVID-specific shock innovations.

#### ELB environment - Non-linear smoothing

This paper implements the ELB as in Hohberger et al. (2019) and Croitorov et al. (2020), following the piecewise linear Kalman filter algorithm by Giovannini et al. (2021). Given the parameter estimates of the model, the algorithm identifies structural shocks that account for endogenous ELB periods using the OccBin approach of Guerrieri and Iacoviello (2015). Thus, it generates a sequence of smoothed variables and shocks that are consistent with the occasionally binding constraint.<sup>11</sup> The sequence of regimes for Germany, i.e., non-binding or binding ELB, is reported in Table B.5 in Appendix B.

 $<sup>^{11}\</sup>mathrm{See}$  Giovannini et al. (2021) and Cardani et al. (2023) for a detailed description of the methodological details.

#### 6. Dynamic transmission of shocks

This section examines the estimated dynamic effects of shocks characteristic of the COVID-19 pandemic and recovery period from 2020 to 2023. Accordingly, it compares the dynamic transmissions between scenarios where the ELB on the short-term nominal interest rate is binding (piecewise) and when it is not binding (linear). When the ELB is binding, economic contraction during the COVID-19 period, marked by decreasing output and inflation rates, is not countered by expansionary monetary policy to alleviate the down-turn in economic activity. Conversely, during 2021-23, negative supply-side shocks leading to rising inflation rates and a slowdown in economic activity are further intensified by monetary tightening aimed at reducing inflation.

Therefore, this section presents generalised impulse response functions (GIRFs) for the following shocks: (i) a transitory lockdown shock, (ii) a risk shock in RoW, (iii) a shock to commodity prices (oil), and (iv) supply-side shocks (retail markup and export prices) that capture binding capacity constraints. The GIRFs are considered within a constrained monetary policy setup where short-term interest rates cannot be adjusted. Specifically, following the sequence of regimes in Table B.5, GIRFs with an effective ELB, consistent with the estimated timing and duration of the ELB regime in the model, are obtained by performing the following steps: Set 2020q2 as the starting point - a period in which the ELB is binding for an additional 6 quarters (see Table B.5) - remove the respective estimated shock, and simulate the model with all remaining shocks. Then, perform simulations by adding the respective estimated shock. The difference between the two scenarios provides the GIRFs under the ELB (red-dashed).

All shocks are performed using the estimated shock variance and persistence. Each panel displays the dynamic responses of real GDP, the policy rate, CPI inflation, private consumption, private investment, total hours worked, the real interest rate, the real effective exchange rate (REER), and the trade balance-to-GDP ratio. Real variables are presented in percent deviations from their respective steady states, whereas the policy rate (annualised), CPI inflation (annualised), and the trade balance-to-GDP ratio are expressed as deviations from the steady state in basis points and percentage points, respectively.

#### Transitory lockdown shock

Figure 2 illustrates the dynamic responses to the transitory lockdown shock, which accounts for the majority of the GDP and consumption contraction during the lockdown in the second quarter of 2020. Based on the estimated magnitude of this non-persistent saving shock and the constrained ELB, it reduces private consumption and real GDP by 8.5% and 3.6%, respectively, for one quarter (2020q2). The reduction in consumption demand, coupled



Figure 2: Dynamic responses to a lockdown shock (forced saving).

Note: The trade balance (normalised by GDP), inflation (p.a.), and real interest rate (p.a.) responses are expressed as percentage point, the policy rate (p.a.) as basis point, deviations from steady state. All other responses are percent deviations from steady state. The size of the shock corresponds to the estimated shock variance.

with the absence of monetary stimulus, also decreases investment demand in the short term (red-dashed line). This effect mirrors a risk or uncertainty shock, where consumption and investment move together. The drop in GDP leads to a temporary decline in employment (hours worked), though this decline is less pronounced due to labour adjustment costs. In the short run, the trade balance improves because of reduced domestic and import demand. The impact on CPI inflation is relatively minor, attributed to the transitory nature of the shock and price stickiness. Additionally, the lockdown shock encompasses both demand-side disturbances (precautionary behaviour of households) and supply-side disturbances (enforced business closures).

#### Risk shock in RoW

Figure 3 displays the dynamic transmission for a negative foreign demand shock. This shock is modelled as an additional disturbance to household time preference - a temporary increase in savings - during the COVID-19 pandemic, which decreases foreign consumption,



Figure 3: Dynamic responses to a risk shock in RoW.

Note: The trade balance (normalised by GDP), inflation (p.a.), and real interest rate (p.a.) responses are expressed as percentage point, the policy rate (p.a.) as basis point, deviations from steady state. All other responses are percent deviations from steady state. The size of the shock corresponds to the estimated shock variance.

output, and prices. Unlike the lockdown shock in Figure 2, this additional savings shock exhibits habit persistence, resulting in a sluggish adjustment process. Regarding spillovers to the German economy, the shock induces a 0.4% decline in real GDP in the absence of monetary accommodation. This decline mainly results from the appreciation of the real effective exchange rate (REER) and its negative impact on the trade balance due to reduced import demand from RoW. The decrease in economic activity also reduces employment, real wages, and consequently, domestic consumption. The rise in real interest rates triggers an immediate decline in investment demand (compared to the linear case). Given the ELB constraint during 2020-2021, the spillover from RoW mimics the dynamic effects of an additional risk shock for Germany.

#### A rise in commodity prices

Figure 4 shows the macroeconomic response to an increase in oil prices. Based on the estimated shock size, the higher oil price elevates both the prices of domestically manufac-



Figure 4: Dynamic responses to an increase in commodity prices.

Note: The trade balance (normalised by GDP), inflation (p.a.), and real interest rate (p.a.) responses are expressed as percentage point, the policy rate (p.a.) as basis point, deviations from steady state. All other responses are percent deviations from steady state. The size of the shock corresponds to the estimated shock variance.

tured goods and CPI inflation, with an impact of up to 1.8 percentage points due to the energy component in final consumption demand. The dynamic effects heavily depend on the response of monetary policy. Under active monetary policy, an increase in interest rates leads to higher real rates in the medium term, which dampens domestic demand (investment) and output. The trade balance deteriorates due to higher commodity import prices and REER depreciation. In the case of passive monetary policy, i.e. a delayed monetary tightening due to the ELB, a decline in real rates mitigates the negative effects on investment. Additionally, lower real wages are compensated by an increase in hours worked, resulting in positive short-term effects on GDP.

#### Supply-side disruptions

Figure 5 illustrates the linear dynamic responses (IRFs) to two exemplary supply shocks characteristic of the COVID-19 recovery period from 2021 to 2023: (i) a retail price markup shock (represented by solid lines), and (ii) a permanent increase in DE export prices (rep-



Figure 5: Dynamic responses to supply-side shocks (Retail markup and export price).

Note: The trade balance (normalised by GDP), inflation (p.a.), and real interest rate (p.a.) responses are expressed as percentage point, the policy rate (p.a.) as basis point, deviations from steady state. All other responses are percent deviations from steady state. The size of the shocks correspond to the estimated shock variances.

resented by dashed lines). The former constitutes a wedge between production costs and consumer prices. This can be interpreted as a consumption-specific price markup by re-tailers.<sup>12</sup> Figure 5 focuses on linear IRFs as both shocks are key drivers of GDP and CPI inflation for 2022-23, when the ECB had already started raising interest rates from the ELB during that period.

Both shocks can be viewed as proxies for binding capacity constraints, capturing the dynamics of the Global Supply Chain Pressure Index by Benigno et al. (2022) and mimicking the effects of supply chain disruptions. Both dampen domestic economic activity (consumption and investment); however, only the retail price markup shock significantly increases CPI inflation. Consequently, the shocks differ in their inflation-output trade-off. While the retail

<sup>&</sup>lt;sup>12</sup>The shock,  $At^{p^c}$ , is a negative productivity shock for the final consumption good, C, as described in Equation 23. A similar shock impacting inflation dynamics in the Euro Area has been identified by Cardani et al. (2023).

markup shock raises CPI inflation by almost 2 percentage points, its negative impact on real GDP is smaller (up to 0.4%) compared to the export price shock, which reduces real GDP by up to 0.7% after one year.

#### 7. Estimated drivers of macroeconomic variables

Historical shock decompositions (SDs) attribute the dynamics of endogenous (observed) model variables to different (groups of) exogenous shocks. This process provides insight into the relative importance of exogenous factors in driving economic fluctuations. This subsection offers a quantitative evaluation of the main estimated drivers of Germany's historical data pattern during the period 2008q1-2023q4.<sup>13</sup> The assessment considers the endogenous ELB periods obtained from non-linear smoothing, as discussed in subsection 5. The approach for extending standard linear (additive) historical shock decomposition to account for occasionally binding constraints is detailed in Appendix C.<sup>14</sup>

Figures 6 - 8 exhibit the SDs of real GDP growth, CPI inflation, and nominal wage growth in Germany. In each subplot, the continuous black line represents the historical data, from which the steady state (1.25% p.a. for GDP and 2.0% p.a. for inflation) has been subtracted. The vertical red bars illustrate the contribution of each respective (group of) exogenous shocks to the data, ensuring that the sum of all exogenous drivers matches the historical data. Given the substantial number of shocks, we categorise them as follows: (1) domestic supply shocks, which include productivity, price, and wage markup shocks; (2) domestic demand shocks, encompassing private savings and investment risk premiums; (3) monetary policy shocks; (4) fiscal policy shocks; (5) commodity price shocks, including oil and industrial supplies; (6) international trade shocks, involving preferences for domestically produced versus foreign goods, export and import price markups, and exchange rate shocks; (7) world demand and supply shocks, containing foreign (REA and RoW) demand and supply shocks; (8) COVID-specific shocks, covering lockdown and labour market shocks; and (9) initial values.<sup>15</sup>

#### Real GDP growth

Comparing the GFC recession in 2008-09 with the COVID-19 pandemic, Figure 6 reveals

<sup>&</sup>lt;sup>13</sup>Focusing on the GFC and COVID-19 crisis periods, the SDs illustrate the reduced sample size from 2008 to 2023. In Figures B.1 and B.2 of Appendix B, the historical drivers of real GDP growth and inflation are shown over the full sample period, using the same shock groupings as discussed in this section.

<sup>&</sup>lt;sup>14</sup>A similar methodology for adapted piecewise linear smoothed shock decomposition is presented in Croitorov et al. (2020) and Cardani et al. (2023). The latter also provides methodological details in their Appendix.

 $<sup>^{15}</sup>$ Figures 6 - 8 do not display the initial values as their impact is minimal given the estimation's initialisation in 1999.



#### Figure 6: Historical decomposition of real GDP growth (yoy)

Note: Real GDP growth is shown in percentage-point deviations from steady state, which is calibrated to 1.25% per year. 0.01 on the y-axis corresponds to 1 pp.

significant differences in the dynamic patterns of Germany's real GDP growth. The recession during the GFC proved to be more persistent compared to the COVID-specific growth volatility in 2020, which was primarily driven by the imposed lockdown and containment measures. The main drivers of these two crisis periods also differ substantially. During the GFC, lower domestic and world demand were major contributors to the recession, whereas monetary policy easing and, to a lesser extent, fiscal expansion played a positive role in mitigating the recession's severity. The normalisation of demand and investment conditions led to a relatively quick recovery in 2010.

The COVID-19 recession, on the other hand, is characterised by both demand and supply disturbances. The model highlights the significant impact of imposed lockdown shocks in DE and the REA, particularly due to the transitory (forced) savings shock, which caused a drop in domestic consumption, and increased investment risk. These factors together accounted for an 8 percentage point (pp) contraction in 2020q2 from the trend. Given the global ramifications of COVID-19, spillovers from world demand and supply shocks also contributed to the decline in real GDP growth in 2020, averaging 1 pp. The positive contribution of international trade during 2020 stems from the reduction in German import demand from REA and RoW, which positively affects GDP growth through increased net exports. Discretionary fiscal policy played a crucial role in stabilising GDP growth in 2020, contributing up to 2 pp on average. The stabilising impact of the government stimulus packages aligns with the findings of Funke and Terasa (2022) and Hinterlang et al. (2023).

The subsequent recovery phase during the pandemic years 2021-23 is characterised by counterbalancing factors. The normalisation of domestic demand (including the easing of lockdown and containment measures) and the recovery of world demand contributed positively to real GDP growth. However, a slowdown in international trade and negative supply shocks acted as downward pressures, keeping growth rates moderately low. Regarding international trade, the model identifies a significant export price shock, reflecting increasing supply-chain bottlenecks that negatively impacted economic activity in Germany by up to 4 pp in 2021-22. The moderate positive contributions from monetary policy, despite the monetary tightening (raising interest rates) in 2022, arise because our model-implied policy rate was higher than the observed policy rate. This gap was closed by negative (easing) monetary policy shocks, implying that the model would have suggested raising interest rates more aggressively or starting earlier.<sup>16</sup> However, monetary policy contributes negatively to GDP growth in 2023.

#### CPI inflation

The drivers of Germany's consumer price inflation (CPI) differ markedly from those influencing real GDP growth, particularly during the COVID-19 pandemic. Figure 7 illustrates that transitory lockdown shocks (within the COVID-19 shock group) have a very marginal impact on current inflation during the pandemic and the subsequent recovery. Despite their significant negative effects on GDP, these shocks do not affect inflation expectations.

The inflation dip during the GFC in 2008-09 was primarily driven by demand-side factors,

 $<sup>^{16}\</sup>mathrm{It}$  should be noted that 'monetary policy' refers only to Taylor-rule shocks and excludes the tapering and reversal of unconventional measures.



#### Figure 7: Historical decomposition of CPI inflation (yoy)

Note: CPI inflation is shown in percentage-point deviations from steady state, which is calibrated to 2% per year. 0.01 on the y-axis corresponds to 1 pp.

including domestic and world demand as well as financial shocks such as increased investment risk, and the appreciation of the euro (international trade). In contrast, Figure 7 highlights that the surge in CPI inflation in 2021-22 was significantly influenced by rising commodity prices and supply-side factors. The increase in energy prices alone contributed up to 4.5 pp to CPI inflation in 2022, accounting for roughly two-thirds of the inflation surge. The reversal of persistent negative global demand shocks indicates a recovery from their pre-COVID trends. Expansionary fiscal policy has played a moderately positive role in driving inflation. However, the VAT reduction during the third and fourth quarters of 2020 had a significant impact on the inflation dip, accounting for around 1.5 pp.

In 2023, decreasing commodity prices provided a downward pressure on inflation by lowering production and consumption costs. However, this effect has been counterbalanced by consumption-specific price markup shocks imposed by retailers (see Section 6). These markup shocks in the retail sector, manifesting as increased import price markups, reflect binding capacity constraints and increasing supply chain disruptions for the German economy.<sup>17</sup> This markup shock contributed up to 4 pp to above-trend inflation in 2023. The moderate positive contributions from monetary policy to CPI inflation in 2022 once again reflect the cautious approach to monetary tightening compared to our model-implied policy rule.

Overall, and in contrast to the GFC, the SDs of CPI inflation align with the findings of Menz (2024), which indicate that the inflation surge during 2021-22 was predominantly driven by supply-side factors.

#### Nominal wage growth (compensation per hour)

Regarding wage developments in Germany, pandemic-related support measures created a significant disparity between compensation per employee (CPE) and compensation per hour (CPH).<sup>18</sup> The volatility in hours worked, driven by job retention schemes - particularly short-time work (STW) schemes in Germany - significantly contributed to this disparity. Since the government's compensation for wage losses is paid directly to employees as social transfers, while wages and salaries decrease in proportion to the reduced hours worked, STW schemes cause a substantial downward shift in CPE. Consequently, as CPE growth declined significantly at the start of the pandemic in 2020, CPH increased due to the substantial reduction in hours worked.<sup>19</sup>

Figure 8 visualises the effects of STW schemes through the COVID-specific labour hoarding shock. The reduction in hours worked contributed positively to wage growth (per hour) in 2020 by around 4 pp (COVID-19 shock group), with adverse effects in 2021, partly due to base effects from 2020. Since the model does not include labour hoarding shocks during the GFC period in 2009, the wedge between CPE and CPH during that time is explained by

 $<sup>^{17}</sup>$ Section 8 discusses and compares estimated shocks to off-model indicators, such as the Global Supply Chain Pressure Index by Benigno et al. (2022).

<sup>&</sup>lt;sup>18</sup>The negative correlation between hours worked and compensation per hour due to job retention schemes was also observed during the GFC recession, though to a much lesser extent than during the pandemic.

<sup>&</sup>lt;sup>19</sup>Bodnár et al. (2023) and Da Silva et al. (2020) provide detailed discussions on wage developments in the euro area and short-time work schemes, respectively, during the pandemic.



Note: Nominal wage growth, measured as compensation per hour, is shown in percentage-point deviations from steady state, which is calibrated to 2.9% per year. 0.01 on the y-axis corresponds to 1 pp.

labour demand and wage-markup shocks (domestic supply).<sup>20</sup> The main drivers of belowtrend wage growth in Germany during the period from 2012 to 2019 were primarily low domestic and world demand, whereas euro depreciation and international trade components contributed positively. The lockdown-related drop in domestic demand was the main

<sup>&</sup>lt;sup>20</sup>The estimated labour hoarding shock during the COVID-19 period can also serve as a proxy for labour market tightness, which is typically measured as the ratio of vacancies (v) to unemployed individuals (u). The contribution of labour market tightness as discussed in Menz (2024) is similar to this finding.

negative driver during 2020. In contrast, the normalisation and recovery of domestic and foreign demand contributed positively to wage growth during the pandemic recovery period from 2022 to 2023. These positive drivers during the recovery period were partly offset by domestic supply-side components, such as retail import price markups, and capacity constraints, which were captured by export price shocks and international trade components.

#### 8. Comparison of shocks to off-model indicators

This section compares the pattern of estimated shocks to off-model indicators that are not included in the observed data, in order to assess the empirical plausibility of the identified shocks. Figure 9 illustrates the comparison between the estimated shocks to the investment risk premium, the COVID-specific lockdown shock, and the retail import price markup shock to their real-world counterparts. All the shocks are central to the historical decomposition of real GDP growth and inflation dynamics in Germany during the COVID-19 pandemic period.

The shock to investment risk in the model represents a wedge between actual investment and the investment demand compatible with average financing costs and expected future returns to investment. This can be interpreted, for example, as an increase in corporate financing costs over safe interest rates or quantitative restrictions on credit volumes. Figure 9a shows that the profile of the estimated shocks to the investment risk premium closely tracks the indicator of credit tightness in the German economy.<sup>21</sup> The investment risk premium (solid line) fell prior to 2008 and rose sharply during the GFC. It then dropped significantly after the financial crisis but increased again during the sovereign debt crisis starting in 2012. It has steadily but slowly declined towards pre-GFC levels, before rising once more during the COVID-19 pandemic. Figure 9a illustrates that the estimated investment risk premia align well with periods of sharp credit tightening during the financial crisis. The estimated premium also matches the easing of credit conditions after the sovereign debt crisis and the subsequent tightening during the pandemic.

Figure 9b compares the pattern of the estimated lockdown shock to alternative pandemicrelated restriction indicators for Germany: Google's mobility indicator and the Oxford COVID-19 Government Response Tracker (Hale et al., 2021). The latter captures government policies concerning closures, containment, and health and economic measures. The estimated lockdown shock shows a strong correspondence with both indicators. The slight

<sup>&</sup>lt;sup>21</sup>Credit tightening as a performance indicator is based on the ECB bank survey on the tightening in bank lending standards (the percentage of banks reporting a tightening of lending standards minus the percentage of banks reporting a loosening of standards). Credit tightness at a given date is measured by the cumulative net tightening in preceding periods.



Figure 9: Off-model indicators and estimated model shocks.

(c) Retail import price markup and GSCPI

Note: In Panel a), the model shock (blue solid line) corresponds to the estimated investment risk premium as described in Equation A.4. The data on credit tightening are sourced from the ECB's bank lending survey. In Panel b), the model shock (blue solid line) represents the estimated lockdown shock as described in Equation 3. The data used are time series for Google's mobility indicator (red dotted line) and the Oxford stringency index (Hale et al., 2021) (yellow dashed line), both of which are scaled to their maximum absolute values. In Panel c), the model shock (blue solid line) represents the estimated price markup shock by import retailers. The data for the Global Supply Chain Pressure Index (GSCPI) are taken from the NY Fed website based on Benigno et al. (2022). Data on credit tightness and the GSCPI have been adjusted using a linear transformation to align the mean and standard deviation of the adjusted series with the corresponding moments of the estimated model shocks.

decoupling from the stringency indicator after the first lockdown (2020q2) suggests that private consumption recovered more quickly than the relaxation of restrictions, potentially due to the increased adoption of online retail. Both the alternative indicators and the estimated lockdown shock also align during the subsequent improvement in the epidemiological situation in 2021.

Figure 9c compares the pattern of the estimated retail import price markup shock to the Global Supply Chain Pressure Index (GSCPI) by Benigno et al. (2022), an indicator that measures pressures within the global supply chain and provides a metric for potential supply chain disruptions, such as increasing global transportation costs or delivery times. The estimated price markup by retailers closely matches the pattern of the GSCPI indicator. Notably, during the pandemic in 2020-22, the estimated price markup rose sharply in tandem with increasing global supply constraints, before both indicated reduced pressures at the supply chain towards the end of 2022 and the beginning of 2023. Figure 9c also shows that the estimated retail price markup shocks were well-aligned in the aftermath of the GFC.

#### 9. Counterfactual analysis without COVID-specific shocks

This section provides a counterfactual analysis to evaluate the benefits of incorporating COVID-specific shocks during the pandemic period. Specifically, it illustrates how these heteroskedastic shocks impact the model fit and the economic interpretation of the COVID-19 period. Figure 10 compares the shock decomposition of real GDP growth, CPI inflation, and wage growth during 2018q1 - 2023q4 between two model variants: (i) the model with COVID-specific shocks and (ii) a version without these shocks. The latter is generated by running a smoother with identical estimated parameters, but with all COVID-specific shocks deactivated. The groups of exogenous shocks are similar, except that the former model variant visualises all COVID-related shocks in a separate group (light blue). Note that Figures 10a, 10c, and 10e provide the identical historical decomposition as in Figures 6-8, respectively.

The corresponding Figures without COVID-specific shocks (10b, 10d, 10f) highlight significant differences during the 2020-21 period. As discussed in Section 7, the real GDP drop in 2020q2 in the baseline model primarily attributes the decline to a transitory (forced) savings shock, as it can better fit the large contraction and subsequent recovery implied by the lockdown measures (10a). In contrast, the counterfactual model, which excludes COVIDspecific shocks, attributes the drop to standard (persistent) savings shocks (10b). However, the more sluggish consumption response in this model affects the path of future expectations, resulting in: (i) a stronger negative demand impact on CPI inflation in 2020 (d), and (ii) more pronounced offsetting supply-side factors, such as price and wage markup shocks, to match the observed time series. For nominal wage growth (compensation per hour), positive wage markup shocks (negative supply shocks) are needed in 2020 to reconcile the decline of hours worked in the absence of COVID-related labour hoarding shocks (10f).<sup>22</sup>

 $<sup>^{22}</sup>$ In an early ex-ante assessment of the COVID-19 pandemic, Mckibbin and Fernando (2020) model the initial phase of the pandemic via preference and risk shocks to predict the adjustment without explicitly incorporating lockdown-related shocks. This approach is comparable to Figure 10b, which provides a reasonable economic interpretation. The added value of transitory (lockdown) shocks lies in their marginal impact



Figure 10: Counterfactual without COVID-specific shocks.

Note: CPI inflation is shown in percentage-point deviations from steady state, which is calibrated to 2% per year. 0.01 on the y-axis corresponds to 1 pp.

In summary, in the absence of COVID-specific shocks, the model requires higher shock variances to account for the observed data patterns, particularly during 2020. This suggests that the adjustment dynamics implied by the model with COVID-specific shocks more closely

on the path of future expectations and, consequently, the improved fit of nominal variables.

align with the observed data during the COVID-19 period, indicating a superior model fit.

#### 10. Conclusion

This paper estimates a state-of-the-art three-region DSGE model to analyse the macroeconomic drivers of the COVID-19 pandemic and recovery period in Germany. By incorporating COVID-specific shocks and the use of commodities in the model, the results suggest a central role for lockdown shocks via forced savings in explaining the contraction of economic activity in Germany during 2020, with significant stabilising effects from fiscal policy measures. Global demand and supply shocks also affected Germany's economic conditions during the pandemic.

The GDP recovery from 2021 to 2023 is characterised by the balancing effects of domestic and foreign demand normalisation against a slowdown in international trade and the exacerbation of supply-chain bottlenecks. The surge in CPI inflation during 2021-22 is primarily attributed to rising commodity prices and domestic and foreign supply-side factors, which mimic the effects of increasing supply chain disruptions. The increase in energy prices alone contributed up to 4.5 pp to CPI inflation in 2022, accounting for roughly two-thirds of the inflation surge. The normalisation of commodity prices in 2023 is counterbalanced by consumption-specific price markups by retailers, creating a wedge between production costs and consumer prices, and consequently slowing the decline in inflation rates in Germany in 2023. The analysis emphasises that supply-side factors were the predominant drivers of inflation during the pandemic recovery period.

The model's estimation results, including shocks to investment risk, consumption-specific lockdown shocks, and price markups by import retailers align closely with off-model indicators, validating the empirical robustness of the identified shocks. This comprehensive analysis provides a plausible narrative for the macroeconomic developments in Germany. It emphasises the importance of considering the variety of domestic and global factors, as well as demand and supply dynamics, in understanding the significant economic fluctuations and heightened volatility caused by the COVID-19 pandemic.

#### References

- ADJEMIAN, S., M. JUILLARD, F. KARAMÉ, W. MUTSCHLER, J. PFEIFER, M. RATTO, N. RION, AND S. VILLEMOT (2024): "Dynare: Reference Manual, Version 6," Dynare Working Papers 80, CEPREMAP.
- ADOLFSON, M., S. LASÉEN, J. LINDÉ, AND M. VILLANI (2008): "Evaluating an estimated new Keynesian small open economy model," *Journal of Economic Dynamics and Control*, 32, 2690–2721.
- ALBONICO, A., L. CALÉS, R. CARDANI, O. CROITOROV, F. FERRONI, M. GIOVANNINI, S. HOHBERGER, B. PATARACCHIA, F. M. PERICOLI, AND R. RACIBORSKI (2019): "Comparing post-crisis dynamics across Euro Area countries with the Global Multicountry model," *Economic Modelling*, 81, 242–273.
- ARCE, O., M. CICCARELLI, C. MONTES-GALDÓN, AND A. KORNPROBST (2024): "What caused the euro area post-pandemic inflation?" Occasional Paper Series 343, European Central Bank.
- BALL, L. M., M. D. LEIGH, AND M. P. MISHRA (2022): "Understanding U.S. Inflation During the COVID Era," IMF Working Papers 2022/208, International Monetary Fund.
- BAYER, C., B. BORN, R. LUETTICKE, AND G. J. MÜLLER (2023): "The Coronavirus Stimulus Package: How Large is the Transfer Multiplier," *The Economic Journal*, 133, 1318–1347.
- BENIGNO, G., J. DI GIOVANNI, J. J. GROEN, AND A. I. NOBLE (2022): "The GSCPI: A new barometer of global supply chain pressures," *FRB of New York Staff Report*.
- BENIGNO, P. AND G. B. EGGERTSSON (2023): "It's Baaack: The Surge in Inflation in the 2020s and the Return of the Non-Linear Phillips Curve," NBER Working Papers 31197, National Bureau of Economic Research, Inc.
- (2024): "The Slanted-L Phillips Curve," AEA Papers and Proceedings, 114, 84–89.
- BLANCHARD, O. AND J. GALÍ (2007): "Real Wage Rigidities and the New Keynesian Model," Journal of Money, Credit and Banking, 39, 35–65.
- BLANCHARD, O. J. AND B. S. BERNANKE (2023): "What Caused the US Pandemic-Era Inflation?" NBER Working Papers 31417, National Bureau of Economic Research, Inc.

- BODNÁR, K., E. GONÇALVES, L. GÓRNICKA, AND G. KOESTER (2023): "Wage developments and their determinants since the start of the pandemic," *Economic Bulletin Articles*, 8.
- BUSSIÈRE, M., G. CALLEGARI, F. GHIRONI, G. SESTIERI, AND N. YAMANO (2013): "Estimating Trade Elasticities: Demand Composition and the Trade Collapse of 2008-2009," *American Economic Journal: Macroeconomics*, 5, 118–151.
- CALÉS, L., F. PERICOLI, AND M. RATTO (2017): "Rotated Slice Sampling: applications for the Bayesian Estimation of DSGE models," mimeo, European Commission.
- CARDANI, R., O. CROITOROV, M. GIOVANNINI, P. PFEIFFER, M. RATTO, AND L. VOGEL (2022a): "The euro area's pandemic recession: A DSGE-based interpretation," *Journal of Economic Dynamics and Control*, 143.
- CARDANI, R., S. HOHBERGER, P. PFEIFFER, AND L. VOGEL (2022b): "Domestic versus foreign drivers of trade (im)balances: How robust is evidence from estimated DSGE models?" *Journal of International Money and Finance*, 121.
- CARDANI, R., P. PFEIFFER, M. RATTO, AND L. VOGEL (2023): "The COVID-19 recession on both sides of the Atlantic: A model-based comparison," *European Economic Review*, 158.
- CHEN, W., M. NEGRO, S. GOYAL, AND A. JOHNSON (2020): "The New York Fed DSGE Model Forecast—December 2020," *Liberty Street Economics December*, 23.
- COMIN, D. A., R. C. JOHNSON, AND C. J. JONES (2023): "Supply Chain Constraints and Inflation," NBER Working Papers 31179, National Bureau of Economic Research, Inc.
- CORRADO, L., S. GRASSI, AND A. PAOLILLO (2021): "Modelling and Estimating Large Macroeconomic Shocks During the Pandemic," CREATES Research Papers 2021-08, Department of Economics and Business Economics, Aarhus University.
- CROITOROV, O., M. GIOVANNINI, S. HOHBERGER, M. RATTO, AND L. VOGEL (2020): "Financial spillover and global risk in a multi-region model of the world economy," *Journal* of Economic Behavior & Organization, 177, 185–218.
- DA SILVA, A. D., M. DOSSCHE, F. DREHER, C. FORONI, AND G. KOESTER (2020): "Short-time work schemes and their effects on wages and disposable income," *Economic Bulletin Boxes*, 4.

- DI GIOVANNI, J. (2022): "How Much Did Supply Constraints Boost U.S. Inflation?" Liberty Street Economics 20220824, Federal Reserve Bank of New York.
- DI GIOVANNI, J., S. KALEMLI-ÖZCAN, A. SILVA, AND M. A. YILDIRIM (2023): "Quantifying the Inflationary Impact of Fiscal Stimulus under Supply Constraints," *AEA Papers* and Proceedings, 113, 76–80.
- DOLLS, M., C. FUEST, AND A. PEICHL (2012): "Automatic stabilizers and economic crisis: US vs. Europe," *Journal of Public Economics*, 96, 279–294.
- EICHENBAUM, M. S., S. REBELO, AND M. TRABANDT (2021): "The Macroeconomics of Epidemics [Economic activity and the spread of viral diseases: Evidence from high frequency data]," *The Review of Financial Studies*, 34, 5149–5187.
- (2022): "Epidemics in the New Keynesian model," *Journal of Economic Dynamics* and Control, 140.
- FUNKE, M. AND R. TERASA (2022): "Has Germany's temporary VAT rates cut as part of the COVID-19 fiscal stimulus boosted growth?" *Journal of Policy Modeling*, 44, 450–473.
- GADATSCH, N., K. HAUZENBERGER, AND N. STÄHLER (2016): "Fiscal policy during the crisis: A look on Germany and the Euro area with GEAR," *Economic Modelling*, 52, 997–1016.
- GAGLIARDONE, L. AND M. GERTLER (2023): "Oil Prices, Monetary Policy and Inflation Surges," NBER Working Papers 31263, National Bureau of Economic Research, Inc.
- GHOMI, M., S. HURTADO, AND J. M. MONTERO (2024): "Reccent Dynamics of Inflation in Spain. An Analysis Based on the Model by Blanchard and Beranke (2023)," Occasional papers 2404, Bank of Spain.
- GIOVANNINI, M., S. HOHBERGER, R. KOLLMANN, M. RATTO, W. ROEGER, AND L. VO-GEL (2019): "Euro Area and U.S. External Adjustment: The Role of Commodity Prices and Emerging Market Shocks," *Journal of International Money and Finance*, 94, 183–205.
- GIOVANNINI, M., P. PFEIFFER, AND M. RATTO (2021): "Efficient and robust inference of models with occasionally binding constraints," Working Papers 2021-03, Joint Research Centre, European Commission.
- GUERRIERI, L. AND M. IACOVIELLO (2015): "OccBin: A toolkit for solving dynamic models with occasionally binding constraints easily," *Journal of Monetary Economics*, 70, 22–38.

- HALE, T., N. ANGRIST, R. GOLDSZMIDT, B. KIRA, A. PETHERICK, T. PHILLIPS, S. WEBSTER, E. CAMERON-BLAKE, L. HALLAS, S. MAJUMDAR, AND H. TATLOW (2021): "A global panel database of pandemic policies (Oxford COVID-19 Government Response Tracker)," *Nature human behaviour*, 5, 529–538.
- HANSEN, N.-J., F. TOSCANI, AND J. ZHOU (2023): "Euro Area Inflation after the Pandemic and Energy Shock: Import Prices, Profits and Wages," IMF Working Paper 2023/131, International Monetary Fund.
- HARDING, M., J. LINDÉ, AND M. TRABANDT (2023): "Understanding post-COVID inflation dynamics," *Journal of Monetary Economics*, 140, Supplement, S101–S118.
- HINTERLANG, N., S. MOYEN, O. RÖHE, AND N. STÄHLER (2023): "Gauging the effects of the German COVID-19 fiscal stimulus package," *European Economic Review*, 154.
- HOHBERGER, S., R. PRIFTIS, AND L. VOGEL (2019): "The macroeconomic effects of quantitative easing in the euro area: Evidence from an estimated DSGE model," *Journal of Economic Dynamics and Control*, 108.
- HOHBERGER, S., M. RATTO, AND L. VOGEL (2020): "The euro exchange rate and Germany's trade surplus," *International Finance*, 23, 85–103.
- (2023): "The macroeconomic effects of unconventional monetary policy: Comparing euro area and US models with shadow rates," *Economic Modelling*, 127.
- JORDA, O. AND F. NECHIO (2023): "Inflation and wage growth since the pandemic," *European Economic Review*, 156, 104474.
- KOLLMANN, R. (2021): "Effects of Covid-19 on Euro area GDP and inflation: demand vs. supply disturbances," *International Economics and Economic Policy*, 18, 475–492.
- KOLLMANN, R., M. RATTO, W. ROEGER, J. IN'T VELD, AND L. VOGEL (2015): "What drives the German current account? And how does it affect other EU Member States?" *Economic Policy*, 30, 47–93.
- MCKIBBIN, W. AND R. FERNANDO (2020): "Global macroeconomic scenarios of the COVID-19 pandemic," in *COVID ECONOMICS 39*, CEPR Press.
- MENZ, J.-O. (2024): "Sources of post-pandemic inflation in Germany and the euro area: An application of Bernanke and Blanchard (2023)," Technical Paper 02/2024, Deutsche Bundesbank, Frankfurt a. M.

NEAL, R. (2003): "Slice Sampling," The Annals of Statistics, 31, 705–741.

- NERI, S., F. BUSETTI, C. CONFLITTI, F. CORSELLO, D. D. MONACHE, AND A. TAGLIAB-RACCI (2023): "Energy price shocks and inflation in the euro area," Questioni di Economia e Finanza (Occasional Papers) 792, Bank of Italy, Economic Research and International Relations Area.
- PASIMENI, P. (2022): "Supply or Demand, that is the Question: Decomposing Euro Area Inflation," *Intereconomics: Review of European Economic Policy*, 57, 384–393.
- PISANI, M. AND A. TAGLIABRACCI (2024): "What caused the post-pandemic inflation in Italy? An application of Bernanke and Blanchard (2023)," Occasional papers (questioni di economia e finanza) 851, Bank of Italy.
- PLANAS, C., M. RATTO, AND A. ROSSI (2015): "Slice sampling in Bayesian estimation of DSGE models," mimeo, European Commission.
- REIS, R. (2022): "The burst of high inflation in 2021–22: how and why did we get here?" BIS Working Papers 1060, Bank for International Settlements.
- ROTEMBERG, J. J. (1982): "Monopolistic Price Adjustment and Aggregate Output," *Review* of *Economic Studies*, 49, 517–531.
- SCHMITT-GROHE, S. AND M. URIBE (2003): "Closing small open economy models," *Journal* of International Economics, 61, 163–185.
- TAYLOR, J. B. (1993): "Discretion versus policy rules in practice," Carnegie-Rochester Conference Series on Public Policy, 39, 195–214.

### Appendix A. Model description

#### Appendix A.1. Households

The Ricardian households maximise the present value of the expected stream of future utility, subject to equation (5), by choosing the amount of consumption,  $C_{j,t}^s$ , and next period asset holdings,  $B_{j,t}^{rf}$ ,  $B_{j,t}^G$ ,  $S_{j,t}$ ,  $B_{j,t}^W$ . The resulting FOCs are:

$$\lambda_{j,t}^{s} = \left[ C_{j,t}^{s} - \varepsilon_{t}^{tC} - h(C_{t-1}^{s} - \varepsilon_{t-1}^{tC}) \right]^{-\theta}, \tag{A.1}$$

$$1 = \tilde{\beta}_t E_t \left[ \frac{\lambda_{j,t+1}^s}{\lambda_{j,t}^s} \frac{(1+i_t^{rf})}{1+\pi_{t+1}^{C,vat}} \right],$$
(A.2)

$$1 = \tilde{\beta}_t E_t \left[ \frac{\lambda_{j,t+1}^s}{\lambda_{j,t}^s} \frac{(1+i_t^G) - \left(\alpha^{b_0} + \varepsilon_t^B\right)}{1 + \pi_{t+1}^{C,vat}} \right],$$
(A.3)

$$1 = \tilde{\beta}_t E_t \left[ \frac{\lambda_{j,t+1}^s}{\lambda_{j,t}^s} \frac{(1+i_{t+1}^S) - \left(\alpha^{S_0} + \varepsilon_t^S\right)}{1 + \pi_{l,t+1}^{C,vat}} \right],\tag{A.4}$$

$$1 = \tilde{\beta}_{t} E_{t} \left[ \frac{\lambda_{j,t+1}^{s}}{\lambda_{j,t}^{s}} \frac{(1+i_{t}^{W}) \frac{e_{RoW,t+1}}{e_{RoW,t}} - \left(\varepsilon_{t}^{bw} + \alpha^{bw_{0}} + \alpha^{bw_{1}} \frac{e_{RoW,t}NFA_{t}}{P_{t}^{Y}Y_{t}}\right)}{1 + \pi_{t+1}^{C,vat}} \right],$$
(A.5)

where,  $\alpha^{bw1} \frac{e_{RoW,t}NFA_t}{P_t^Y Y_t}$  captures a debt-dependent country risk premium on net foreign asset holdings as external closure to ensure long-run stability (see Schmitt-Grohe and Uribe, 2003; Adolfson et al., 2008).

The optimality conditions are similar to standard Euler equations, but incorporate assetspecific risk premia which depend on exogenous shocks  $\varepsilon_{kt}^B$ ,  $\varepsilon_{kt}^S$ ,  $\varepsilon^b w_{kt}$ . Combining the Euler equation for the risk-free bond (A.2) with (A.3), (A.4) and (A.5), we obtain the following approximated expressions:

$$i_t^G = i_t^{rf} + rprem_t^G, (A.6)$$

$$i_t^S = i_t^{rf} + rprem_t^S, \tag{A.7}$$

$$E_t \left[ \frac{e_{RoW,t+1}}{e_{RoW,t}} \right] i_t^W = i_t^{rf} + rprem_t^W, \tag{A.8}$$

where  $rprem_t^G$  and  $rprem_t^W$  are risk premia on domestic government bonds and foreign bonds, respectively,  $rprem_{l,t}^S$  are the country-specific risk premia on domestic and foreign shares, and  $rprem_t^{ra}$  a global financial shock to the risk appetite.

#### Appendix A.2. Firms

Following Rotemberg (1982), firms face quadratic adjustment costs,  $adj_{i,t}$ , measured in terms of production input factors. Specifically, the adjustment costs are associated with the output price,  $P_{i,t}^Y$ , labour input,  $N_{i,t}$ , capital stock and investment,  $I_{i,t}$ , as well as capacity utilisation variation,  $CU_{i,t}$ :

$$adj_{i,t}^{PY} = \sigma^Y \frac{\gamma^P}{2} Y_t \left[ \frac{P_{i,t}^Y}{P_{i,t-1}^Y} - \exp(\bar{\pi}) \right]^2,$$
 (A.9)

$$adj_{i,t}^{N} = \frac{\gamma^{N}}{2} Y_{t} \left[ \frac{N_{i,t}}{N_{i,t-1}} - \exp(g^{pop}) \right]^{2},$$
 (A.10)

$$adj_{i,t}^{I} = \frac{P_{t}^{I}}{P_{t}^{Y}} \left[ \frac{\gamma^{I,1}}{2} K_{t-1} \left( \frac{I_{i,t}}{K_{t-1}} - \delta_{t}^{K} \right)^{2} + \frac{\gamma^{I,2}}{2} \frac{(I_{i,t} - I_{i,t-1} \exp(g^{Y} + g^{P^{I}}))^{2}}{K_{t-1}} \right],$$
(A.11)

$$adj_{i,t}^{CU} = \frac{P_t^I}{P_t^Y} K_{i,t-1}^{tot} \left[ \gamma^{CU,1} (CU_{i,t} - 1) + \frac{\gamma^{CU,2}}{2} (CU_{i,t} - 1)^2 \right],$$
(A.12)

where  $\gamma$ -s capture the degree of adjustment costs,  $\bar{\pi}$ ,  $g^{pop}$ ,  $g^Y$ ,  $g^{P^I}$  are the steady-state growth rates of inflation, population, and country-specific GDP and investment price deflator, respectively.  $\delta_t^K \neq \delta$  is a function of the depreciation rate adjusted for the capital trend in order to have zero adjustment costs on the trend-path.<sup>23</sup>

Given the Lagrange multiplier associated with the technology constraint,  $\mu^y$ , the FOCs with respect to labour, capital, investment, and capacity utilisation are given by:

$$(1 - \tau^K)\frac{W_t}{P_t^Y} = \alpha \left(\mu_t^y - \varepsilon_t^{ND}\right) \frac{Y_t}{N_t - FN} - \frac{\partial adj_t^N}{\partial N_t} + E_t \left[\frac{1 + \pi_{t+1}^Y}{1 + i_{t+1}^s} \frac{\partial adj_{t+1}^N}{\partial N_t}\right], \qquad (A.13)$$

$$Q_{t} = E_{t} \left[ \frac{1 + \pi_{t+1}^{Y} P_{t+1}^{I} P_{t}^{Y}}{1 + i_{t+1}^{s} P_{t+1}^{Y} P_{t}^{I}} \right] \\ \left( \tau^{K} \delta - \frac{\partial a d j_{t}^{CU}}{\partial K_{t-1}} + Q_{t+1} (1 - \delta) + (1 - \alpha) \mu_{t+1}^{Y} \frac{P_{t+1}^{Y} Y_{kt+1}}{P_{t+1}^{I} K_{t}^{tot}} \right) \right], \quad (A.14)$$

 $<sup>\</sup>overline{{}^{23}\text{We specify } \delta_t^K = exp(g^{\bar{Y}} + GAPI0) - (1 - \delta), \text{ where } g^{\bar{Y}} \text{ and } GAPI0 \text{ are the global GDP trend and the investment-specific technology growth, respectively, so that } \frac{I}{K} - \delta^k \neq 0 \text{ along the trend path.}}$ 

$$Q_{t} = \left[1 + \gamma^{I,1} \left(\frac{(I_{t} + \varepsilon_{t}^{tS})}{K_{t-1}} - \delta_{t}^{K}\right) + \gamma^{I,2} \frac{((I_{t} + \varepsilon_{t}^{tS}) - (I_{t-1} + \varepsilon_{t-1}^{tS})\exp(g^{Y} + g^{P^{I}}))}{K_{t-1}}\right] - E_{t} \left[\frac{1 + \pi_{t+1}^{Y}}{1 + i_{t+1}^{s}} \frac{P_{t+1}^{I}}{P_{t+1}^{Y}} \frac{P_{t}^{Y}}{P_{t}^{I}}\exp(g^{Y} + g^{P^{I}}) + (I_{t} + \varepsilon_{t}^{tS})\exp(g^{Y} + g^{P^{I}})\right] - \gamma^{I,2} \frac{((I_{t+1} + \varepsilon_{t+1}^{tS}) - (I_{t} + \varepsilon_{t}^{tS})\exp(g^{Y} + g^{P^{I}}))}{K_{t}}\right],$$
(A.15)

$$\mu_t^y (1-\alpha) \frac{Y_t}{CU_t} \frac{P_t^Y}{P_t^I} = K_{t-1}^{tot} \Big[ \gamma^{u,1} + \gamma^{u,2} (CU_t - 1) \Big],$$
(A.16)

where  $Q_t = \mu_t / \frac{P_t^I}{P_t^Y}$  represents Tobin's Q and  $Actr_t Pop_t$  is the active labour force of the domestic country. Equation (A.13) characterises the optimal level of labour input, taking into account labour overhead. Equation (A.14) and (A.15) define the Tobin's Q, which is equal to the replacement cost of capital (the relative price of capital).  $\varepsilon_t^{tS}$  is the investment-specific lockdown shock. Finally, (A.16) describes capacity utilisation, where the left-hand side indicates the additional output produced while the right-hand side captures the costs of higher utilisation rate.

Given the Rotemberg set-up and imposing the price symmetry condition,  $P_{i,t}^Y = P_t^Y$ , the FOC with respect to  $P_{i,t}^Y$  yields the New Keynesian Phillips curve:

$$\mu_t^y \sigma^Y = (1 - \tau^K) (\sigma^Y - 1) + \sigma^Y \gamma^P \frac{P_t^Y}{P_{t-1}^Y} \left( \pi_t^Y - \bar{\pi} \right) - \sigma^Y \gamma^P E_t \left[ \frac{1 + \pi_{t+1}^Y}{1 + i_{t+1}^s} \frac{P_{t+1}^Y}{P_t^Y} \frac{Y_{t+1}}{Y_t} \left( \pi_{t+1}^Y - \bar{\pi} \right) \right] + \sigma^Y \varepsilon_t^{\mu Y},$$
(A.17)

where  $\varepsilon_t^{\mu Y}$  is the inverse of the markup shock.

# Appendix B. Additional results

		DE	REA	RoW
Preferences				
Tedendaria and Sector	0	0.000	0.008	0.000
Intertemporal discount factor	р s	0.998	0.998	0.998
Savers share	$\omega^*$	0.01	1.00	1.00
Degree of openness	$\omega_{\alpha M}$	2.5	-	-
Import share in consumption	$^{S}_{cM,C}$	0.30	0.28	0.00
Import share in investment	$s^{M,I}$	0.22	0.17	0.05
Import share in investment	s,-	0.31	-	-
Import share in government expenditure	. M. X	0.51	-	-
Defenses for import from DEA	s M.R.E.A	0.20	0.51	0.15
Preference for imports from <b>D</b> eW	oM,RoW	0.55	-	0.07
Preference for imports from ROW	M.DE	0.23	0.77	-
Preference for imports from DE	8-11,22	-	0.52	0.48
Production				
Cobb-Douglas labour share	α	0.65	1.00	1.00
Depreciation of private capital stock	d V	0.014	-	-
Elasticity of substitution between differentiated goods	$\sigma^{I}$	6.50	-	-
Share of commodities in total output	s <sup>CU</sup>	0.06	0.04	0.05
Linear capacity utilisation adj. costs	$\gamma^{CU,1}$	0.03	-	-
Fiscal policy				
Communities to a	-C	0.00		
Consumption tax	$\tau^{\circ}_{K}$	0.20	-	-
Corporate pront tax	$\tau^{n}$	0.20	-	-
Labour tax Definit towart (in $\%$ of CDP)	$T^{-1}$	0.41	-	-
Debt target (in % of GDP)	$\bar{B}^{G}$	61.6	-	-
Steady-state ratios				
			0.60	0.79
Private consumption share	C/Y	0.55	0.68	0.72
Private consumption share Private investment share	C/Y I/Y	$0.55 \\ 0.19$	0.68 -	-
Private consumption share Private investment share Government consumption share	C/Y I/Y G/Y	$0.55 \\ 0.19 \\ 0.19$	- -	-
Private consumption share Private investment share Government consumption share Government investment share	C/Y I/Y G/Y IG/Y	$0.55 \\ 0.19 \\ 0.19 \\ 0.02$		
Private consumption share Private investment share Government consumption share Government investment share Transfer share	C/Y I/Y G/Y IG/Y T/Y	0.55 0.19 0.19 0.02 0.17		
Private consumption share Private investment share Government consumption share Government investment share Transfer share Trade balance share	C/Y I/Y G/Y IG/Y T/Y TB/Y	0.55 0.19 0.02 0.17 0.04	-0.02	

Table B.1: Selected calibrated structural parameters.

		Prior d	istribution	Posterior d	listribution
		Distr	Mean St.Dev	REA	RoW
	Mor	ietary Po	olicy		
Interest rate persistence	$a^i$	C	0.95	0.01	0.05
interest fate persistence	$\rho$	G	0.85	(0.88, 0.02)	(0.04.0.06)
Response to inflation	$n^{i,\phi}$	G	2.00	(0.88, 0.93)	(0.94, 0.90)
Response to initation	"	G	2.00	(1.56, 2.10)	(1.47, 1.01)
Besponse to CDP	$n^{i,y}$	G	0.20	(1.50, 2.15)	0.07
	"	u	0.04	(0.03, 0.11)	(0.05, 0.15)
	Р	reference	es		
Consumption habit persistence	h	В	0.50 0.20	0.73 (0.72, 0.83)	0.87
Bisk aversion	θ	G	1.50	1.50	1 29
	0	4	0.20	$(1 \ 19 \ 1 \ 71)$	$(1 \ 18 \ 1 \ 72)$
Phillips curve coefficient	$\phi^{Y}$	G	0.025	0.04	0.06
	7	-	0.01	(0.02, 0.05)	(0.03, 0.07)
Import price elasticity	$\sigma^{z}$	G	2.00	3.13	1.28
r · · · · · · · · · · · · · · · · · · ·		-	0.40	(2.62, 3.88)	(1.11, 1.43)
Oil price elasticity	$\sigma^{o}$	G	0.5	0.31	0.12
			0.1	(0.30, 0.35)	(0.01, 0.29)

Table B.2: Prior and posterior distribution of estimated model parameters in REA and RoW.

Note: Cols. (1)-(2) list model parameters. Cols. (3)-(4) indicate the prior distribution function (B: Beta distribution; G: Gamma distribution). Col. (5)-(6) show the mode and the 90% HPD intervals of the posterior distributions of REA and RoW model parameters.

		Prior dis	stribution	Posterior distribution
		Distr	Mean St.Dev	DE
Autoco	rrelations	s of forcing	g variable:	5
Subjective discount factor	$\rho^{UC}$	Beta	0.50	0.83
			0.20	(0.75, 0.90)
Investment risk premium	$\rho^S$	Beta	0.85	0.94
			0.05	(0.88, 0.94)
Labour demand	$\rho^{ND}$	Beta	0.50	0.78
	.,		0.20	(0.74, 0.83)
Trade share	$\rho^M$	Beta	0.50	0.91
	~		0.20	(0.88, 0.95)
Government consumption	$\rho^G$	Beta	0.50	0.97
	_		0.20	(0.95, 0.98)
Government transfers	$\rho^T$	Beta	0.50	0.94
	~ ~		0.20	(0.90, 0.95)
Commodity imports	$\rho^{CO}$	Beta	0.50	0.82
			0.20	(0.76, 0.86)
Productivity growth	$\rho^{GA}$	Beta	0.50	0.82
			0.20	(0.94, 0.98)
International bond preference	$\rho_{EA}^{BW}$	Beta	0.50	0.87
			0.20	(0.71, 0.89)

#### Table B.3: Selected estimated exogenous shock processes for DE.

Standard deviations (%) of innovations to forcing variables

Subjective discount factor	$\varepsilon^{UC}$	Gamma	1.00	1.74
			0.40	(0.64, 2.50)
Investment risk premium	$\epsilon^{S}$	Gamma	0.10	0.25
			0.04	(0.20, 0.44)
Price mark-up	$\varepsilon^{MUY}$	Gamma	2.00	8.77
			0.80	(5.39, 9.80)
Labour demand	$\epsilon^{ND}$	Gamma	1.00	2.81
			0.40	(2.56, 2.94)
Trade share	$\varepsilon^{M}$	Gamma	1.00	1.94
			0.40	(1.74, 2.05)
International bond preference	$\varepsilon^{BW}_{EA}$	Gamma	1.00	0.21
			0.40	(0.15, 0.40)
Labour supply	$\varepsilon^U$	Gamma	1.00	2.73
			0.40	(1.94, 2.88)
Export price	$\varepsilon^{PX}$	Gamma	1.00	0.49
			0.40	(0.39, 0.59)
Government consumption	$\varepsilon^{G}$	Gamma	1.00	0.19
			0.40	(0.16, 0.22)
Government transfers	$\varepsilon^T$	Gamma	1.00	0.15
			0.40	(0.13, 0.17)
Commodity imports	$\varepsilon^{CO}$	Gamma	1.00	5.05
			0.40	(4.02, 5.86)
Productivity growth	$\varepsilon^{GA}$	Gamma	0.10	0.05
			0.04	(0.02, 0.06)
Productivity trend	$\varepsilon^A$	Gamma	0.10	0.02
			0.04	(0.01, 0.05)
Monetary policy	$\varepsilon^{i}_{EA}$	Gamma	1.00	0.10
			0.40	(0.08, 0.11)

Note: Cols. (1)-(2) list model parameters. Cols. (3)-(4) indicate the prior distribution function (B: Beta distribution; G: Gamma distribution). Cols. (5) shows the mode and the 90% HPD intervals of the posterior distributions.

		Prior dis	tribution	Posterior d	listribution
		Distr	Mean St.Dev	REA	RoW
Aut	ocorrelat	ions of fo	rcing varia	bles	
Subjective discount factor	$\rho^{UC}$	Beta	0.50	0.69	0.79
	v		0.20	(0.59, 0.74)	(0.73, 0.87)
Price mark-up	$\rho^{1}$	Beta	0.50	0.57	0.62
Trade share	$\rho^M$	Beta	0.20	(0.33, 0.63) 0.94 (0.90, 0.96)	(0.52, 0.69) 0.97 (0.95, 0.99)
Commodity imports	0 <sup>CO</sup>	Beta	0.20	0.90	-
	F		0.20	(0.85, 0.96)	
Productivity growth	$\rho^{GA}$	Beta	0.50	0.93	0.94
			0.20	(0.90, 0.95)	(0.91, 0.95)

Table B.4: Selected estimated exogenous shock processes for REA and RoW.

 $\varepsilon^{UC}$ Subjective discount factor Gamma 1.001.950.690.40(1.53, 2.69)(0.46, 1.01) $\varepsilon^{MUY}$ Price mark-up Gamma 1.000.170.43 $(0.14, \ 0.28)$  $(0.34, \ 0.53)$ 0.40 $\varepsilon^{M}$ Trade share  $\operatorname{Gamma}$ 1.003.142.89(2.53, 3.16)0.40(2.69, 3.42) $\varepsilon^{CO}$ Commodity imports Gamma 1.004.86-0.40(4.02, 5.86) $\varepsilon^{GA}$ Productivity growth Gamma 0.10 0.03 0.08 (0.03, 0.04)(0.07, 0.10)0.04 $\varepsilon^i$ Monetary policy  $\operatorname{Gamma}$ 1.000.100.090.40(0.08, 0.11)(0.08, 0.11)

Note: Cols. (1)-(2) list model parameters. Cols. (3)-(4) indicate the prior distribution function (B: Beta distribution; G: Gamma distribution). Cols. (5) shows the mode and the 90% HPD intervals of the posterior distributions.

time	regime sequence	starting period of regime
2008Q1	0	1
2008Q2	0	1
2008Q3	0	1
2008Q4	0	1
2009Q1	0 1 0	1 8 11
2009Q2	0 1 0	159
2009Q3	010	147
2009Q4	0	1
2010Q1	0	1
2010Q2	0	1
2010Q0 2010Q4	0	1
2011Q1	0	1
2011Q2	0	1
2011Q3	0	1
2011Q4	0	1
2012Q1	0	1
2012Q2	0	1
2012Q3	0	1
2012Q4	0	1
2013Q1	0	1
2013Q2	010	146
2013Q3	010	136
2013Q4	010	145
2014Q2	010	136
2014Q3	010	126
2014Q4	1 0	1 6
2015Q1	1 0	1 6
2015Q2	1 0	1 5
2015Q3	1 0	1 5
2015Q4	1 0	1 6
2016Q1	1 0	1 7
2016Q2	10	16
2016Q3	10	15
2010Q4	10	1 3
2017Q1	10	13
2017Q3	1 0	1 3
2017Q4	1 0	1 3
2018Q1	1 0	1 3
2018Q2	1 0	1 3
2018Q3	1 0	1 3
2018Q4	1 0	1 3
2019Q1	1 0	1 3
2019Q2	10	1 3
2019Q3	10	13
2019Q4	10	13
2020Q1	10	1.6
2020@2	10	15
2020Q4	1 0	1 5
2021Q1	1 0	1 3
2021Q2	1 0	1 3
2021Q3	1 0	1 3
2021Q4	1 0	1 3
2022Q1	1 0	$1 \ 2$
2022Q2	0	1
2022Q3	0	1
2022Q4	U	1
2023Q1	0	1
202302	0	1
2023Q4	0	1

Table B.5: Historical sequence of occasionally binding regimes.

Note: First column: [0] unconstrained; [0 1 0] indicates an unconstrained regime, but agents expect to be binding in the future; [1 0] indicates a constrained regime. Second column: [1 8 11] indicates an expected constrained regime starting in 8 periods ahead and last for additional 3 periods; [1 6] indicates a constrained regime with an expected duration of additional 6 periods.



Figure B.1: Historical decomposition of real GDP growth (yoy)

Note: Real GDP growth is shown in percentage-point deviations from steady state, which is calibrated to 1.25% per year. 0.01 on the y-axis corresponds to 1 pp.



Figure B.2: Historical decomposition of CPI inflation (yoy)

Note: CPI inflation is shown in percentage-point deviations from steady state, which is calibrated to 2% per year. 0.01 on the y-axis corresponds to 1 pp.

#### Appendix C. Piecewise linear smoothed shock decomposition

This appendix describes the extension of the standard linear (additive) historical shock decomposition to occasionally binding constraints. The piecewise linear smoother provides an estimate of the historical sequence of regimes, i.e. identifies periods in which the ELB has been binding (Table B.5). The sequence of regimes triggers a sequence of state space matrices:

$$y_t = C(t) + T(t)y_{t-1} + R(t)\epsilon_t,$$

where y are endogenous variables in deviation from steady state,  $\varepsilon$  are the smoothed shocks, and C(t) is a constant which is triggered by the ELB regime. While C = 0 in normal times, the constant is triggered by the fact that, under the ELB regime, the Taylor-rule becomes  $i_{EA} = i_{EA}^{LB}$  and violates the steady state solution as  $i_{EA}^{LB} < \overline{i}$ .

The smoothed shock decomposition is performed similarly to the usual linear case: Given the smoothed series of regimes, the shocks are propagated individually through the sequence of state space matrices T and R. The array C is, instead, treated as an additional 'exogenous' process, the so-called 'regime effect'. The regime effect results from the interaction of all shocks hitting the system simultaneously, for all times  $\langle = t$ . Hence, we can assume that such a regime effect is also a function of the model shocks, which allows us to compute, at each time point and for each  $y_j$  of interest, the absolute value of the contribution of each shock  $\varepsilon_i$  onto variable  $y_j$ :

$$w_{j,i}(t) = |y_j(e_i, t)|,$$

which provides a set of weights that can be used to apportion the regime affect among all shocks in the model. The intuition behind this procedure is the following: If a shock is relevant for y at a given point in time t, it will also be relevant in triggering the regime effect. For example, if it is an expansionary shock, it would contribute to mitigate the duration of the constrained regime and vice versa. By doing so, we obtain a historical shock decomposition in terms of the usual model shocks, which also includes the regime effect.

#### Appendix D. Data source and transformations

#### Data sources

Data for the Germany and EA (quarterly national accounts, fiscal aggregates, quarterly interest and exchange rates) are taken from Eurostat. DE and EA imports of commodities from RoW are based on BEA data and on Eurostat Comext data. RoW series are constructed on the basis of the IMF International Financial Statistics (IFS) and World Economic Outlook (WEO) databases.

#### Data series for RoW variables

Series for GDP, investment, prices and interest rates in RoW starting in 1999 are constructed on the basis of data for the following 58 countries: Albania, Algeria, Argentina, Armenia, Australia, Azerbaijan, Belarus, Brazil, Bulgaria, Canada, Chile, China, Colombia, Croatia, Czech Republic, Denmark, Egypt, Georgia, Hong Kong, Hungary, Iceland, India, Indonesia, Iran, Israel, Japan, Jordan, Korea, Lebanon, Libya, FYR Macedonia, Malaysia, Mexico, Moldova, Montenegro, Morocco, New Zealand, Nigeria, Norway, Philippines, Poland, Romania, Russia, Saudi Arabia, Serbia, Singapore, South Africa, Sweden, Switzerland, Syria, Taiwan, Thailand, Tunisia, Turkey, Ukraine, United Arab Emirates, United Kingdom, and Venezuela. The RoW data are annual data from the IMF International Financial Statistics (IFS) and World Economic Outlook (WEO) databases.

#### List of observables

The estimation uses the time series information for 41 endogenous variables. We additionally observe the first quarter of the capital stock and the net international investment position to initialise the starting point. Table D.6 lists the observed time series. We apply logarithmic transformations to all observables, with the exception of the trade balance-to-GDP ratio, the oil price (Brent), the commodities import share to GDP, the price of commodities imports, and nominal interest rates. Figure D.3 plots the observed DE data pattern for the estimation. GDP deflators and relative prices of demand components are computed as the ratios of the current-price value to the chain-indexed volume series. Note that we observe EA aggregate variables and compute model-consistent REA variables given the size of Germany. Table D.6: Observed times series.

DE	EA	RoW
GDP (nominal and real)	GDP (nominal and real)	GDP (nominal and real)
TFP trend	GDP trend	GDP trend
Private consumption (nominal and real)	Interest rate (nominal)	Interest rate (nominal)
Total investment (nominal and real)	Effective exchange rate (nominal)	Oil price (Brent) in USI
Hours worked	Exports (nominal and real)	Population
Wages (nominal)	Imports (nominal and real)	
Exports (nominal and real)	Commodities import share (nominal)	
Imports (nominal and real)	Price of commodity imports (from RoW)	
Government debt (nominal)	Population	
Gov. consumption (nominal and real)		
Gov. investment (nominal)		
Government transfers (nominal)		
Gov. interest payments (nominal)		
Commodities import share (nominal)		
Price of commodity imports (from RoW)		
Active population rate		
Population		



